

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Höckel, Lisa Sofie

Working Paper Speaking the same language: The effect of foreign origin teachers on students' language skills

Ruhr Economic Papers, No. 862

Provided in Cooperation with: RWI – Leibniz-Institut für Wirtschaftsforschung, Essen

Suggested Citation: Höckel, Lisa Sofie (2020) : Speaking the same language: The effect of foreign origin teachers on students' language skills, Ruhr Economic Papers, No. 862, ISBN 978-3-86788-999-5, RWI - Leibniz-Institut für Wirtschaftsforschung, Essen, https://doi.org/10.4419/86788999

This Version is available at: https://hdl.handle.net/10419/222907

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



RUHR ECONOMIC PAPERS

Lisa Sofie Höckel



CWI #862

Imprint

Ruhr Economic Papers

Published by

RWI – Leibniz-Institut für Wirtschaftsforschung Hohenzollernstr. 1-3, 45128 Essen, Germany Ruhr-Universität Bochum (RUB), Department of Economics Universitätsstr. 150, 44801 Bochum, Germany Technische Universität Dortmund, Department of Economic and Social Sciences Vogelpothsweg 87, 44227 Dortmund, Germany Universität Duisburg-Essen, Department of Economics Universitätsstr. 12, 45117 Essen, Germany

Editors

Prof. Dr. Thomas K. Bauer RUB, Department of Economics, Empirical Economics Phone: +49 (0) 234/3 22 83 41, e-mail: thomas.bauer@rub.de Prof. Dr. Wolfgang Leininger Technische Universität Dortmund, Department of Economic and Social Sciences Economics - Microeconomics Phone: +49 (0) 231/7 55-3297, e-mail: W.Leininger@tu-dortmund.de Prof. Dr. Volker Clausen University of Duisburg-Essen, Department of Economics International Economics Phone: +49 (0) 201/1 83-3655, e-mail: vclausen@vwl.uni-due.de Prof. Dr. Ronald Bachmann, Prof. Dr. Roland Döhrn, Prof. Dr. Manuel Frondel, Prof. Dr. Ansgar Wübker RWI, Phone: +49 (0) 201/81 49-213, e-mail: presse@rwi-essen.de

Editorial Office

Sabine Weiler

RWI, Phone: +49 (0) 201/81 49-213, e-mail: sabine.weiler@rwi-essen.de

Ruhr Economic Papers #862

Responsible Editor: Ronald Bachmann All rights reserved. Essen, Germany, 2020

ISSN 1864-4872 (online) - ISBN 978-3-86788-999-5

The working papers published in the series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editors.

Ruhr Economic Papers #862

Lisa Sofie Höckel

Speaking the Same Language – The Effect of Foreign Origin Teachers on Students' Language Skills



Bibliografische Informationen der Deutschen Nationalbibliothek

The Deutsche Nationalbibliothek lists this publication in the Deutsche Nationalbibliografie; detailed bibliographic data are available on the Internet at http://dnb.dnb.de

RWI is funded by the Federal Government and the federal state of North Rhine-Westphalia.

http://dx.doi.org/10.4419/86788999 ISSN 1864-4872 (online) ISBN 978-3-86788-999-5 Lisa Sofie Höckel¹

Speaking the Same Language – The Effect of Foreign Origin Teachers on Students' Language Skills

Abstract

This study investigates whether foreign origin teachers causally affect their students language skills in secondary school. Exploring within-student variation in assignment to teachers, I find that teachers who are immigrants or descendants of immigrants significantly increase the reading comprehension of students. The effect is strongest for foreign origin students and can partly be explained by a role model effect. The positive effect of bilingual foreign origin teachers is particularly large. They significantly increase reading comprehension for native and foreign origin students. Given their own experience in language learning, bilingual teachers seem exceptionally well-equipped in teaching languages to both native and foreign origin students.

JEL-Code: J13, J15, I24

Keywords: Academic achievement; education economics; language skills; matching; migration; role model effect; teacher bias effect

August 2020

¹ Lisa Sofie Höckel, RWI and IZA. – I am grateful to Ronald Bachmann, Thomas K. Bauer, Julia Bredtmann, Jan Feld, Colin Green, Carla Haelermans, Ingo Isphording, David Jaeger, Fernanda Martínez Flores, Manuel Santos Silva, Marcus Tamm, Christina Vonnahme and Matthias Westphal and participants at the AEA 2019 Annual Meeting, the IZA Annual Migration Meeting, the IZA Summer School in Labor Economics, the InternationalWorkshop on Applied Economics of Education, the EALE, the Verein für Socialpolitic Annual Meeting as well as seminar participants at RWI, the WZB Berlin and Rutgers University for valuable comments and suggestions. All remaining errors are my own. – This paper uses data from the National Educational Panel Study (NEPS): Starting Cohort Grade 5, doi:10.5157/NEPS:SC3:9.0.0. From 2008 to 2013, NEPS data were collected as part of the Framework Program for the Promotion of Empirical Educational Research funded by the German Federal Ministry of Education and Research (BMBF). As of 2014, NEPS is carried out by the Leibniz Institute for Educational Trajectories (LIfBi) at the University of Bamberg in cooperation with a nationwide network. - All correspondence to: Lisa Sofie Höckel, RWI, Hohenzollernstr. 1-3, 45128 Essen, Germany, e-mail: lisa.hoeckel@rwi-essen.de

1 Introduction

International migration not only affects the labor force, it also increases diversity in classrooms around the world. In OECD countries, more than 25 percent of 15-34 year-olds report a foreign origin, and the share of foreign origin students continues to rise (OECD, 2018). Large and persistent achievement gaps between native and foreign origin students, meaning students who are either immigrants or descendants of immigrants, prevail in most countries (e.g., Schnepf, 2007; Algan *et al.*, 2010), and the economics literature has identified language as the single most important determinant associated with the achievement gap (e.g., Dustmann *et al.*, 2010, 2012; Geay *et al.*, 2013). Increasingly, policy makers consider the achievement gap as one of the most pressing problems of the education system with severe implications for income and other lifelong outcomes of foreign origin students (OECD, 2018).

Among teachers, those of foreign origin are often underrepresented. In the US, immigrants make up approximately 13 percent of the population, but they comprise only 11 percent of teachers (Furuya *et al.*, 2019). Considering second and higher generation immigrants, this gap even widens. In Germany, for example, whereas 33 percent of under-15 year-old students have a foreign origin, this is only true for 8 percent of primary and secondary school teachers (Statistisches Bundesamt, 2012, 2018). Therefore, policy makers promote the "fast and sustainable" recruitment of foreign origin teachers to facilitate the integration of foreign origin students (Bundesamt für Migration und Flüchtlinge, 2010). Yet, we know little about the effect of foreign origin teachers on students' academic achievements. Are foreign origin teachers better equipped in helping foreign origin children to overcome potential language barriers and socio-economic disadvantages? And how do they affect the achievement of native students? For the US, the literature shows that minority students benefit from same-race teachers (e.g., Dee, 2004; Fairlie *et al.*, 2014; Gershenson *et al.*, 2016), but it is unclear if these findings can be transferred to the context of immigration in Europe.

In this paper, I investigate whether having a foreign origin teacher causally affects

students' language skills in lower secondary school, holding constant both observed and unobserved factors related to academic outcomes. Using data from the German National Educational Panel Study (NEPS) with unique information on teachers' foreign origin and their teacher examination grade and exploiting within-student variation in assignment to teachers, I find that foreign origin teachers increase objective reading comprehension of their students. The effect is strongest for foreign origin students and can partly be explained by a positive role model effect. More specifically, foreign origin students perceive their foreign origin teachers more favorably. Foreign origin teachers who report a mother tongue other than German increase reading test scores for both native and foreign origin students the most. Ruling out alternative explanations and finding no effect of bilingual teachers on math test scores, I therefore argue that bilingual teachers are particularly well-equipped for teaching languages.

Theoretically, foreign origin teachers might affect students' academic achievement differently than native teachers for two reasons. First, *student-teacher matching* can influence students' academic achievement. Matching effects comprise two complementary channels that can make demographic matching of students and teachers particularly advantageous for matched students. A *role model effect* describes a positive reaction of foreign origin students to foreign origin teachers. Triggered by the teacher's presence, rather than an explicit behavior, foreign origin students' beliefs about their educational possibilities can enhance, making them more enthusiastic, confident, and engaged in class. A *teacher bias effect* refers to the teacher's behavior. Foreign origin teachers can display origin-specific patterns of behavior, they can allocate more class time to interacting with students of the same origin, they can prepare class material more oriented towards same-origin students, or they can simply favor them through grading (Dee, 2007).

Second, foreign origin teachers can have different *language skills* than native teachers. On the one hand, foreign origin teachers can be less proficient in the language of instruction. On the other hand, they can communicate course content to foreign origin students in a language or manner these students are more familiar with (Seah, 2018b). Given potential *student-teacher matching* and *language skills*, the effect of having a foreign origin teacher on students' language skills is theoretically ambiguous.

Previous research has discussed matching effects with respect to demographic characteristics, mostly gender and race (e.g., Dee, 2004; Bettinger and Long, 2005; Hoffmann and Oreopoulos, 2009; Fairlie *et al.*, 2014; Antecol *et al.*, 2015). Dee (2004) examines test scores from the Project STAR class-size experiment, which randomly matches students and teachers within participating schools. He shows that assignment to own-race teachers significantly increases math and reading achievement of both black and white students. Fairlie *et al.* (2014) and Egalite *et al.* (2015) confirm these ethnicity/race matching effects using large administrative data. More recently, Seah (2018a) investigates the effect of immigrant teachers on 8th graders in the US and finds no adverse effect of immigrant teachers on the achievement of (native) students. However, the National Education Longitudinal Study of 1988 (NELS) data employed only allow him to use variation in student attainment across subjects.¹

Empirically, role model and teacher bias effects are difficult to disentangle. Therefore, some studies directly test for teacher bias and discrimination effects. Dee (2005) exploits student-specific evaluations from teachers and shows that ethnic matching between student and teacher has large effects on teachers' perception of student performance. Similarly, Gershenson *et al.* (2016) find that non-black teachers have significantly lower educational expectations for black students, and studies by Hinnerich *et al.* (2015) and Alesina *et al.* (2018) show that immigrant students are graded significantly worse by their teacher in comparison to the blind grading of an external evaluator. Evidence from Germany finds teacher discrimination for essay grades for students with a Turkish-sounding first name (Sprietsma, 2013) and grade penalties in primary school for second generation immigrants (Kiss, 2013). Employing Chilean data to investigate gender matching effects on academic achievement, the study by Paredes (2014) uniquely tests for role model and teacher bias effects in the same setting. She finds that girls benefit from having female teachers and argues that her results are explained by a role model rather than a teacher bias effect.

¹Another limitation of the NELS data is that they do not provide information on the immigrant status of the teacher. Instead, Seah (2018a) defines immigrant teacher as a teacher who reports being a native speaker of a language other than English.

More specifically, she shows that the effect is only significant for subjects with lower proportions of female teachers and for girls with less educated mothers.

A small strand of the literature discusses language skills of teachers. Early studies analyze the effect of foreign teaching assistants on the academic achievements of undergraduate students in university (Borjas, 2000; Asano, 2008). The ambiguous effects found by these studies can be explained by the non-random assignment of teaching assistants to students. In the study most closely related to this paper, Seah (2018b) examines the effect of having a linguistically similar teacher on the academic achievements of secondary school students in the United States. Using data from the NELS, he exploits within-student variation in test scores and the native language of teachers across two subjects. He finds no effect of being assigned to a linguistically similar teacher once the teacher's ethnicity is controlled for. In related work, Chin *et al.* (2013) evaluate the effect of a bilingual education program on the achievement of limited English proficient (LEP) students and their classmates. Employing a regression-discontinuity design, they find no impact on the achievement of students for whom the program was designed (LEP students), but estimate a positive effect for their classmates.

This study contributes to the literature in three ways. First, it adds to the scant evidence on the causal relationship between teachers' foreign origin and students' academic achievement. Given the large and persistent achievement gap between native and foreign origin students, this fills an important gap in the literature. Second, it contributes to the discussion on teacher bias versus role model effects. By exploiting information on objective test scores and subjective teacher grading as well as survey questions on students' perception of their teacher, I can uniquely test both effects in the same setting. Third, it investigates the particular role of bilingual teachers in the context of language acquisition, which has not been studied previously. My results point to a special ability of teachers who report a mother tongue other than German in teaching language skills to students.

The remainder of the study is organized as follows. Section 2 gives a short institutional overview of the German school system, and Section 3 introduces the empirical strategy. Section 4 discusses the data, and Section 5 presents the findings. Section 6 concludes.

2 Institutional Background

A key feature of the German education system is that students are typically tracked after four years of elementary schooling.² Students are sorted based on their academic ability and assigned to one of three secondary school tracks: lower-secondary track (*Hauptschule*), middle-secondary track (*Realschule*) and upper-secondary track (*Gymnasium*).³ *Hauptschule* provides practical education and prepares students for vocational education (until grade 9); *Realschule* has a broader range of emphasis for intermediate students (until grade 10); and *Gymnasium* qualifies students for higher education (until grade 12 or 13).⁴ Depending on the federal state, the track is determined by parental choice or a binding teacher recommendation based on the students' academic achievement and ability to work independently. Schooling is compulsory for nine or ten years, depending on the federal state. Students usually finish the track they have been assigned to, but switching tracks is possible and became more common in recent years.

A second relevant feature of the German education system is that teaching is organized in classes rather than courses. More specifically, class refers to a group of up to 30 students who are allocated by their school's headmaster to the same classroom upon entry into secondary school. All students in one class share the same teacher for a given subject. In contrast to the US, students do not take different courses in the same subject based on their proficiency.⁵ Further, teachers do not specialize in teaching a particular grade but the school's headmaster assigns them to classes on a yearly basis. Typically, the composition of students in a class rarely changes throughout the school years.

Most German teachers are graduates of a formal teaching education program (*Lehramt-studium*). Conditional on having earned a degree that qualifies for tertiary education, teacher training for secondary education comprises two components. First, teacher candidates complete four to six years of university courses covering the two subjects they later

²In the federal states of Berlin and Brandenburg, students are tracked after six years of schooling.

³Additionally, there are comprehensive schools (*Gesamtsschulen*) that combine all education types and amount to 12 percent of all German secondary schools (Malecki *et al.*, 2014).

 $^{^4\}mathrm{See}$ Figure A1 for an illustration of the German education system.

⁵This study focuses on German language classes that are compulsory in German schools. Students are obliged to take them and cannot choose courses based on their preferences for teachers.

want to teach in combination with some pedagogical training. At the end of the first phase, the candidate takes exams on pedagogic and theoretical knowledge of the subjects studied. In addition to the grades earned at the university level, these exams comprise the first state examination grade. Second, candidates participate for 18-24 months in a practical program of teaching seminars (*Referendariat*) at a teaching training school. During this phase, candidates have teaching positions, complete a thesis, and deliver demonstration lessons rated by head teachers. The combination of assessments of the demonstration lessons, the thesis grade and exams sum up to the second state examination.

The second state examination is compulsory for entry into civil service.⁶ Candidates who have studied abroad can only become civil servants by exhibiting comparable educational attainment, or by obtaining the second state examination. Ultimately, the grade of the second state examination, in combination with the local demand for teachers, determines the school a teacher is assigned to. Federal states can hire teaching candidates without the second state examination on regular salaried positions without awarding them civil servant status.

3 Empirical Strategy

An ideal empirical setting to study the effect of having a foreign origin teacher on students' academic achievement requires random allocation of teachers across classes. Otherwise, simple OLS regressions might lead to biased estimates due to two main threats to identification: First, native and foreign origin teachers might select into schools with students who systematically differ with respect to their proficiency. In most German federal states, teachers are allocated to schools partly based on their second state examination grade. Additionally, residential sorting by socio-economic status produces significant quality differences between school catchment areas even within small geographical areas (Noreisch, 2006). If foreign origin teachers were better teachers and had better grades, they could, for example, be more likely to be sent to schools with better performing students.

⁶In most federal states, teachers become civil servants. Nevertheless, approximately one in four teachers is hired as an employee rather than a civil servant.

Consequently, this allocation would lead to an overestimation of the effect of foreign origin teachers.

Second, within schools, the headmaster could allocate teachers to more or less proficient classes based on their origin or confounding factors correlated with origin. For example, foreign origin teachers could be allocated to classes with a high share of foreign origin students, and, thus more heterogeneous classes. If class heterogeneity is negatively correlated with student performance, the effect of foreign origin teachers would be underestimated.

In order to address these biases, I follow the teacher value-added literature, which decomposes students' test scores into components related to student heterogeneity and components related to teacher quality (e.g., Rothstein, 2010). More specifically, I use longitudinal data with class fixed effects to estimate the "gain" in student achievement if a foreign origin teacher teaches a student using variation in teacher assignment over time.

For the class fixed effects specification, I estimate the model

$$y_{iclt} = \beta FT_{ct} + \phi' X_{it} + \gamma' C_l + \rho_c + \delta_t + \varepsilon_{iclt}$$

$$\tag{1}$$

where y_{iclt} is the outcome of student *i* in class *c* with teacher *l* in year *t*. FT_{ct} is a dummy variable for the German language teacher being of foreign origin and X_{it} is a vector of observed student characteristics. C_l denotes a vector of observed teacher characteristics and ρ_c is a class fixed effect. δ_t are year dummies and ε_{iclt} is the error term.

Eq. (1) represents the return to both individual and teacher inputs. The main variable of interest is FT_{ct} , whose parameter β captures the effect of having a foreign origin teacher. Employing class fixed effects allows me to implicitly standardize test scores across classes as within classrooms students experience the same class-level shocks and take exactly the same tests. The key assumption is that there are no unobserved factors in ε_{iclt} that correlate with students' outcomes and teacher assignment to classes.

One major threat to identification is non-random teacher assignment with respect to students' foreign origin.⁷ If foreign origin teachers are assigned to classes where foreign

⁷To test for systematic sorting of teachers to classrooms, I regress student characteristics that are likely to be correlated with unobserved ability on a dummy variable indicating a foreign origin teacher.

origin students perform better relative to native students and this performance is not due to the foreign origin interaction between students and teacher but due to unobserved differences in ability between native and foreign origin students, the estimates would be biased. Therefore, I test for differential sorting by regressing student characteristics that are likely to be correlated with unobserved ability on a foreign origin student-teacher interaction.

Table 1 shows the interaction coefficients of the estimations, which measure how much the native foreign origin gap in outcomes varies across courses taught by native and foreign origin teachers. All specifications are insignificant and most coefficients are close to zero. As math and German test scores are highly correlated, I view column (4) as a particularly good measure of a potential unobserved student characteristic that might be related to a differential selection. Overall, this balancing test gives confidence that there is no systematic sorting with respect to the foreign origin student-teacher interaction.

	(1) Female student		(3) Class repeated	(4) Math testscore
Foreign origin student-teacher interaction	-0.033 (0.052)	2.691 (6.393)	-0.010 (0.012)	-0.007 (0.136)
Class FE	Yes	Yes	Yes	· · · ·
Observations	7,313	6,969	7,351	7,346

Table 1: Student-Teacher Sorting by Foreign Origin

Notes: – Each estimate is from a regression of the corresponding student characteristic on a dummy indicating the teacher's origin, a dummy indicating the student's origin and the interaction between these two variables in a classroom. (Regression results without class FE are comparable and displayed in Table A1.) Robust standard errors are in parentheses. – *** p < 0.01, ** p < 0.05, * p < 0.10.

Further sources of endogeneity could arise from changes in class composition over time and unobserved student characteristics.⁸ Therefore, I adapt Eq. (1) by including student fixed effects and dropping student-level variables from Eq. (1) that are multicollinear with the student fixed effects. This within-student identification strategy uses variation in assignment to teacher due to student mobility and teacher turnover and accounts for

Table A2 presents the results of this balancing test. For gender, age and class repeated, I cannot reject the hypothesis that the correlation between student characteristics and foreign origin teacher is equal to zero. The positive coefficient on math test score can be explained by the higher probability of having a foreign origin teachers to work in schools with an academic tracks (*Gymnasium*).

⁸A related source of endogeneity results from dynamic assignment of teachers into classes (e.g. good teachers follow after bad teachers). While I can not directly test for dynamic selection, it is unlikely to bias the estimates from Eq. (1) as long as the dynamic selection is orthogonal to the foreign origin of teachers.

time-invariant student heterogeneity (e.g., with respect to student's unobserved ability or motivation). This strategy yields unbiased estimators if there are no unobserved, time-varying student-specific factors that are correlated with both students' outcomes and class assignment.⁹

For the student fixed effects specification, I estimate the model

$$y_{iclt} = \beta FT_{ct} + \phi' X_{it} + \gamma' C_l + \rho_c + \delta_t + \omega_i + \varepsilon_{iclt}$$
⁽²⁾

where ω_i are student fixed effects.

4 Data and Descriptive Statistics

This study uses data from the German Educational National Panel Study (NEPS) (Blossfeld, 2011). The NEPS has a multi-cohort design and draws from a representative sampling frame of students from six starting cohorts. The survey follows students as they move through the education system and contains extensive questionnaires answered by persons in the students' personal environment, such as parents, teachers, and headmasters.

For the empirical analysis, I employ data from starting cohort 3 (SC3) as SC3 provides unique information on teachers' origin. SC3 follows students from grade 5 to 9, an age cohort that is particular suitable for the research question as the first nine years of education are compulsory and crucial for lifelong education outcomes (Angrist and Krueger, 1991). The sampling population of SC3 contains all German fifth graders in schools offering lower secondary education in school year 2010/11. First, schools are randomly drawn from the population of public schools to be representative by school type.¹⁰ Second, a class is randomly selected within each school (see Steinhauer and Zinn, 2016, for sampling design). Participation is voluntary and implies that students complete competency tests and answer a questionnaire on their socioeconomic background. The teacher questionnaire contains

⁹This specification controls for absolute sorting that takes place if students who take classes with foreign origin teachers are systematically different from those who do not—irrespective of their foreign origin.

¹⁰Students from schools with a predominant foreign teaching language and students who are not able to participate in the normal testing procedure are excluded.

information on the teachers' demographics and aspects of their career choice and studies. I disregard interview data on parents and headmasters to minimize sample attrition.

In order to determine the students' foreign origin, I employ three variables provided by the NEPS data set. (a.) Students report which citizenship they hold. If they mention any (additional) nationality other than German, I code them as foreign origin. (b.) Interviewers record students' country of origin. I treat non-German groups of origin as foreign origin. (c.) Students state if they have a Russian or Turkish migration background. If they mention one of the two migration backgrounds, I code them as foreign origin.¹¹

To determine German language teachers' foreign origin, I use two variables. (a.) Teachers are asked about their migration background, namely if they are foreign born or if they have at least one foreign born parent. (b.) Teachers report their mother tongue, meaning the language they learned as a child in the family. If they mention a mother tongue additional to German, or other than German, I code them as foreign origin. If teachers do not report a migration background or a mother tongue, I code them as native Germans.

Besides containing unique information on teacher origin, the NEPS has the advantage of providing objective and unidimensional competence scores. For the empirical analysis, I focus on reading comprehension scores as the main outcome variable. The score is designed to measure the ability to understand and use written texts, which are important preconditions to develop personal skills and participate in social life and the labor force (Gehrer *et al.*, 2012). As promoting reading comprehension is one of the key objectives of German language classes, I can attribute this skill to the domain of the German language teacher.

A relevant feature of the competence tests is that NEPS interviewers conduct the test and German language teachers cannot manipulate the score. This procedure is crucial in a setting where teacher bias effects might be at work. Reading comprehension is assessed by multiple choice questionnaires, which test the understanding of five text functions and

¹¹This question does not bias the sample towards students of Russian and Turkish background. It only identifies an additional 6 percent of foreign origin students.

associated text types, namely informational, commenting, instructional, advertising and literary texts. The reading competence test lasts 28 minutes per text function and fits the thematic orientation, lexical, semantic, and grammatical properties of the specific age cohort (Gehrer *et al.*, 2012). The answers to the multiple choice questions are aggregated by a weighted maximum likelihood estimation and constrained to having a mean of zero in the first wave. This standardization ensures that scores are comparable across different survey waves.

I impose several restrictions on the data. From the initially 6,527 students, I observe 6,485 in the years when the tests take place. Of those, I drop 358 students because they do not participate in the competence test. I can link 5,758 students to their German language teacher and I have context data for 719 German language teachers of 4,724 students.¹²

To illustrate the data, Table 2 presents descriptive statistics for the students observed. The first variable describes the main outcome variable, i.e., the objective reading comprehension scores. The average of 0.29 indicates that the reading comprehension in the sample has increased over time.¹³ Out of the 4,724 students in the sample, 35 percent are of foreign origin. This share is in line with statistics of the German Microzensus, which estimate a share of (narrowly defined) foreign origin students of 33 percent in 2012 (Statistisches Bundesamt, 2012). The most common countries of origin in the data are Russia, Turkey, and Poland, which together account for 45 percent of foreign origin students. The sample is balanced with respect to gender, and almost 95 percent of the students are born in 1999 or 2000. The sample contains observations from grade 5, 7 and 9, because reading comprehension is tested in those grades.¹⁴ The grade average of 6.65 indicates that the sample contains more observations in lower grades. This pattern can be explained by sample attrition and 3 percent of repeated grades. The average household size is 4 and around 80 percent of students live in households of 3 to 5 people.

 $^{^{12}}$ To avoid further sample attrition, I keep teacher observations with missing information if the teacher answers at least two questions used for the control variables.

 $^{^{13}}$ Further, the sample is slightly skewed towards higher achieving students as the average reading comprehension in grade 5 is 0.07 rather than 0.

¹⁴The missing years are not necessarily a concern, as teachers typically change classes every 2 years implying that a teacher teaching a certain subject in grade 5 (7) is likely to teach the same class in grade 6 (8).

	Mean	Min	Max	Student Obs.	Panel Obs.
Main outcome variable					
Reading comprehension score	0.285	-4	6	4,724	7,351
Student characteristics				*	,
Female	0.494	0	1	4,724	7,351
Foreign origin	0.352	0	1	4,724	7,351
Birth month					
January	0.082	0	1	4,724	7,351
February	0.078	0	1	4,724	7,351
March	0.084	0	1	4,724	7,351
April	0.079	0	1	4,724	7,351
May	0.078	0	1	4,724	7,351
June	0.085	0	1	4,724	7,351
July	0.090	0	1	4,724	7,351
August	0.092	0	1	4,724	7,351
September	0.092	0	1	4,724	7,351
October	0.081	0	1	4,724	7,351
November	0.080	0	1	4,724	7,351
December	0.079	0	1	4,724	7,351
Birth year					
1995	0.001	0	1	4,724	7,351
1997	0.002	0	1	4,724	7,351
1998	0.045	0	1	4,724	7,351
1999	0.413	0	1	4,724	7,351
2000	0.530	0	1	4,724	7,351
2001	0.010	0	1	4,724	7,351
2002	0.001	0	1	4,724	7,351
Grade	6.647	5	9	4,724	7,351
Grade repeated	0.025	0	1	4,724	7,351
Household size	4.439	1	10	4,724	7,351
Other outcome variables					
Mathematics score	0.304	-5	5	4,720	7,346
German grade	4.370	1	6	4,700	8,572
German teacher tries to understand my point	3.283	1	5	4,584	7,024
German teacher has class under control	3.702	1	5	4,666	7,213

Table 2: Descriptive Statistics – Students

Notes: – The descriptive statistics are weighted to account for differences in the number of observations per student.

Besides reading comprehension scores and student characteristics, Table 2 provides information on further outcome variables. It shows an average mathematics score of 0.30, indicating that the mathematical test score also increases over time. The German grade corresponds to the subjective grade given by the German language teacher at the end of the school year and it ranges from one to six. Here, six corresponds to an outstanding achievement while one (and two) refers to an insufficient achievement. The most prevalent grade is satisfactory (4), which 44 percent of students receive. The NEPS data also contain survey questions on how students perceive their teachers. Students are asked to agree with statements about their German language teacher on a scale ranging from 1 to 5.¹⁵ In the sample, 28 percent of students rather agree with the statement that their teacher tries

¹⁵The categories are: Does not apply at all (1), does rather not apply (2), partly (3), does rather apply (4), applies completely (5).

to understand their point of view and then tells them what he/she would do. 31 percent of students state that their teacher has the class completely under control.

Overall, the characteristics of native and foreign origin students differ significantly with respect to all variables except for gender and birth month (see Table A3). In line with the existing literature, the data reveal a large native-foreign gap in achievement with a mean reading comprehension score for native students of 0.42 (0.47 for math) and 0.06 (0.04 for math) for foreign origin students. In addition, foreign origin students are older and more likely to live in larger households.

	Mean	Min	Max
Foreign origin	0.063	0	1
First generation immigrant	0.022	0	1
Second and higher generation immigrant	0.040	0	1
Bilingual	0.040	0	1
Slavic	0.014	0	1
Romance	0.014	0	1
Others	0.013	0	1
Female	0.752	0	1
Age	44.170	24	66
Birth decade			
1940s	0.031	0	1
1950s	0.292	0	1
1960s	0.208	0	1
1970s	0.228	0	1
1980s	0.241	0	1
Age when profession was chosen			
Between 0-14	0.145	0	1
Between 15-19	0.446	0	1
Between 20-24	0.210	0	1
After 25	0.104	0	1
Grade at first state exam	-0.010	-2	3
German language studies	0.900	0	1
Language studies	0.236	0	1
Teacher-year observations	7,351		
Teacher observations	718		

Table 3: Descriptive Statistics – German Language Teachers

Notes: – The descriptive statistics are weighted to account for differences in the number of observations per teacher.

Table 3 summarizes the main explanatory variable and the teacher characteristics controlled for in the empirical analysis. Of the German language teachers observed in the sample, 6 percent are of foreign origin. This share is comparable with national estimates for 2012 (7.8 percent) (Statistisches Bundesamt, 2018). More than 60 percent of foreign origin teachers are born in Germany, implying that the majority of foreign origin teachers are second and higher generation immigrants. Further, 63 percent of foreign origin teachers report a mother tongue other than (or additional to) German. In contrast to the student population, teachers are evenly distributed across countries of origin and mother tongues. In the sample, 34 percent of foreign origin teachers have a Slavic mother tongue (mainly Polish and Russian), 34 percent a Romance mother tongue (mainly Italian), and 32 percent another (*Other*) mother tongue (mainly English). Less than one in four German language teachers is male, and the German teacher's average age is 44. Besides age and birth decade, I further control for the age when the person decided to become a teacher. Almost 45 percent of teachers report that they chose their profession in their teens.

While the question on age when profession was chosen might be prone to response bias, I include it to control for unobserved intrinsic motivation. To proxy German language teachers' overall ability, I include their average grade at the first state examination which is standardized to have a mean of zero and a standard deviation of one in the full sample. Of the German language teachers in the sample, 90 percent studied German language studies and 24 percent studied other languages such as English or Slavic languages. I include controls for the federal state where the German language teacher acquired his/her higher education entrance qualification, but I do not report them due to data confidentiality.

Native and foreign origin teachers differ significantly with respect to all variables, except the federal state where they acquired their higher education entrance qualification (see Table A4). Foreign origin teachers are younger and decided to become teachers earlier than native teachers. Further, their average grade at the first state examination is 0.17 standard deviations below the grade of native teachers and they are more likely to teach in the academic track (*Gymnasium*). While 25 percent of native teachers work in *Realschule*, only 10 percent of foreign origin teachers do so. In sum, native and foreign origin teachers differ with respect to important observable characteristics. Therefore, it is crucial to control for these characteristics.

Overall, the sample consists of 255 classes (2,911 students) with a teacher transition. Of those, 31 classes (419 students) experience a change in the foreign origin of the teacher. While the sample size is limited, the identifying variation is larger than in previous studies (c.f., Borjas, 2000; Fleisher *et al.*, 2002; Asano, 2008) and comparable to the US study by Seah (2018a).

5 Results

5.1 Foreign Teacher Effect

I start the empirical analysis by testing if foreign origin teachers affect students' reading comprehension. Panel A of Table 4 displays the aggregated foreign origin teacher effect for the full sample. Panels B and C provide separate estimates for native and foreign origin students to investigate if there are differential effects by students' origin. As we move along the columns, I increasingly restrict the variation used to identify the parameter of interest. The specification in column (1) controls for student characteristics and class fixed effects. Following Eq. (1), it accounts for peer effects and teacher allocation across schools and classes. Column (2) includes teacher characteristics, and column (3) displays results from the preferred specification described in Eq. (2), which contains both student and classroom fixed effects.¹⁶

For the overall sample, there is a positive and significant correlation between reading comprehension scores and having a foreign origin teacher in column (1). The foreign origin teacher estimate becomes larger once its effect is disentangled from teacher characteristics in column (2). In column (3), the effect is identified by within-student variation in teacher allocation. Consequently, less variation is employed and β is less precisely estimated. Nevertheless, the effect stays significant and the magnitude rises to 0.26. The coefficient of 0.26 amounts to 0.20 standard deviations in the reading comprehension test score. This effect is sizable in comparison to teacher value-added effects found in the literature. Rockoff (2004), for example, finds an increase in teacher quality by one standard deviation to improve student test scores by 0.1 standard deviations. Other seminal papers of the teacher value-added literature document similar effect sizes (c.f., Rivkin *et al.*, 2005; Aaronson *et al.*, 2007; Chetty *et al.*, 2014a,b).

For native students, in Panel B, the effect size of having a foreign origin teacher is similar in the specifications including class and student fixed effects (column (2) and

¹⁶For ease of exposition, coefficients of the control variables are presented in Table A7 and Table A8. They are in the expected direction.

Panel A: All Students	(1)	(2)	(3)
Foreign origin teacher	0.175^{**}	0.232^{**}	* 0.259*
	(0.069)	(0.069)	(0.137)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.377	0.378	0.660
Observations	$7,\!351$	$7,\!351$	$7,\!351$
Panel B: Native Students			
Foreign origin teacher	0.260**	0.315^{**}	* 0.279
	(0.108)	(0.107)	(0.223)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.362	0.363	0.648
Observations	4,812	4,812	4,812
Panel C: Foreign Origin Students			
Foreign origin teacher	0.115	0.129	0.375^{*}
	(0.087)	(0.095)	(0.198)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.394	0.394	0.655
Observations	2.539	2.539	2.539

Table 4: Effect of Foreign Origin Teacheron Reading Comprehension

(3)). For foreign origin students, in Panel C, the correlation between student unobserved heterogeneity and the teacher's origin matters. With student fixed effects, the effect of having a foreign origin teacher doubles in size and becomes statistically significant. Overall, in the preferred third specification, the foreign origin teacher effect is larger and more precisely estimated for foreign origin than native students. Especially for foreign origin students, this improvement in reading comprehension can have important implications for lifelong outcomes such as productivity and wages because language fluency facilitates the transfer and adaptation of skills in the job market (Chiswick and Miller, 1995; Dustmann and van Soest, 2001; Hayfron, 2001; Dustmann and Fabbri, 2003).

Nevertheless, the effects are positive and comparable in size for both sub-samples.¹⁷ This finding is in line with Seah (2018a), who shows no adverse effects of immigrant

¹⁷Results obtained from a fully interacted model do not reveal significantly different effects of foreign origin teachers on students with and without foreign origin.

teachers on native students in the US. In contrast to the literature on race-matching, the positive effect of foreign origin teachers is not caused by matching effects in the sense that there are positive effects on the matched group (here: foreign origin students) and adverse effects on the mismatched group (here: native students). Instead, Table 4 indicates positive effects for foreign origin students and clearly no negative effect for native students. However, I do not find a universal positive effect of foreign origin teachers on students' academic achievement as the effect of foreign origin teachers on math test scores is indistinguishable from zero (see Table A9). Therefore, the following section focuses on German language teachers and analyzes teacher bias and role model effects as well as language specific skills as potential mechanisms for the positive effect on students' reading comprehension.

5.2 What Explains the Foreign Teacher Effect?

5.2.1 Teacher Bias Effect

The positive effect of foreign origin teachers in Table 4 is stronger for foreign origin students. Therefore, I test if a teacher bias effect, meaning an explicit behavior of teachers targeted towards matched students (e.g., grading them more favorably), can explain the differential effect sizes between native and foreign origin students.

So far, the specification implicitly assumes that foreign origin students can be influenced by any foreign origin teacher equally and by a similar amount. An alternative approach is to allow foreign origin interaction to operate only if a student is matched with a teacher of a similar language group. Therefore, I analyze if teachers are more likely to increase the reading comprehension of students who share a similar origin. Such an effect could indicate that teachers allocate more class time to students whose origin they match with.

In Table 5, I employ a language match variable as the main variable of interest.¹⁸ The language match variable is equal to one if student and teacher report a mother tongue of

¹⁸Using a linguistic distance measure to proxy the match with respect to the disadvantage in reading comprehension arising from the linguistic distance between student and teacher (e.g., Isphording, 2014) yields in similar results (see Table A10).

Panel A: Native Students	(1)	(2)	(3)
Language match	-0.352^{**}	*-0.467**	*-0.533**
	(0.108)	(0.127)	(0.263)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.362	0.363	0.649
Observations	4,812	4,812	4,812
Panel B: Foreign Origin Students			
Panel B: Foreign Origin Students Language match	0.003	-0.022	0.324
Panel B: Foreign Origin Students Language match	0.003 (0.283)	-0.022 (0.356)	0.324 (0.475)
Panel B: Foreign Origin Students Language match Class FE	0.003 (0.283) Yes	-0.022 (0.356) Yes	0.324 (0.475) Yes
Panel B: Foreign Origin Students Language match Class FE Student controls	0.003 (0.283) Yes Yes	-0.022 (0.356) Yes Yes	0.324 (0.475) Yes Yes
Panel B: Foreign Origin Students Language match Class FE Student controls Teacher controls	0.003 (0.283) Yes Yes No	-0.022 (0.356) Yes Yes Yes	0.324 (0.475) Yes Yes Yes
Panel B: Foreign Origin Students Language match Class FE Student controls Teacher controls Student FE	0.003 (0.283) Yes Yes No No	-0.022 (0.356) Yes Yes Yes No	0.324 (0.475) Yes Yes Yes Yes
Panel B: Foreign Origin Students Language match Class FE Student controls Teacher controls Student FE Adjusted R ²	0.003 (0.283) Yes Yes No No 0.394	-0.022 (0.356) Yes Yes Yes No 0.393	0.324 (0.475) Yes Yes Yes Yes 0.653

Table 5: Effect of Language Match between Teacherand Student on Reading Comprehension

the same language group, e.g. a student with German as a mother tongue and a teacher with German as a mother tongue. Notably, there is a negative matching effect for native students. Yet, this negative language match for native students is not surprising given the positive effect of foreign origin teachers who mostly do not match the students' German mother tongue. For foreign origin students, there is no significant positive matching effect. The point estimate of having a student-teacher language match is smaller than the effect of having a foreign origin teacher. Furthermore, the standard errors are larger.¹⁹

As a second test for teacher bias, I analyze the students' German grades. In Germany, few centralized exams are conducted until grade 9, and the grading of students is mostly left to the discretion of the teacher. A positive effect of language matches on subjective teacher grading could suggest a teacher bias. Table 6 uses students' German grades as an outcome variable and shows, in Panel A, a positive correlation between having a language match and the German grade for native students. For foreign origin students, Table 6 does not reveal a positive matching effect.²⁰ Therefore, the results fail to confirm the existence

 $^{^{19}}$ These results are in line with Fairlie *et al.* (2014), who find the effect of same-race and minority interaction to be similar.

²⁰I obtain similar results running regressions on a harmonized sample of students taking the reading comprehension test in Table A11.

of a teacher bias effect for foreign origin teachers.²¹ Instead, the results illustrate that the foreign teacher effect is not driven by specific language matches between foreign origin students and teachers.

Panel A: Native Students	(1)	(2)	(3)
Language match	0.111	0.131^{**}	0.116
	(0.079)	(0.060)	(0.083)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.224	0.223	0.640
Observations	5,757	5,757	5,757
Panel B: Foreign Origin Students			
Language match	-0.370	-0.376	-0.091
	(0.361)	(0.362)	(0.197)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Teacher controls Student FE	No No	Yes No	Yes Yes
Teacher controls Student FE Adjusted R ²	No No 0.174	Yes No 0.175	Yes Yes 0.625

Table 6: Effect of Language Match between Teacherand Student on German Grade

Notes: – Results are obtained from OLS regressions. All reported standard errors are clustered at the teacher level. – *** p < 0.01, ** p < 0.05, * p < 0.10.

5.2.2 Role Model Effect

Teacher bias and role model effects are difficult to disentangle. Studies in a university setting interpret exposure to female faculty members or female instructors in initial courses as female role model effects (e.g., Canes and Rosen, 1995; Bettinger and Long, 2005). Yet, these studies cannot rule out direct teacher influence via teacher bias effects. To overcome this problem, Dee (2007) compares students' perception on the subject taught by matched and unmatched teachers to elicit the role model effect more directly. Likewise, I approximate role mode effects by employing survey questions on how students perceive their teachers. If teacher's foreign origin can explain part of the variation in students' perception of their teacher, this effect can be attributed to a role model effect.²²

 $^{^{21}\}rm Using$ for eign origin teacher instead of the language match variable as the main explanatory variable, Table A12 does not show a teacher bias effect.

²²In this context, I define a role model effect as a positive teacher perception of the student irrespective of this perception being caused by an underlying teacher bias effect or not.

Table 7: Effect of Foreign Origin Teacher on Students' Perceptionthat the Teacher Understands Their Point

Panel A: Native Students	(1)	(2)	(3)
Foreign origin teacher	-0.243^{**}	-0.153	-0.064
	(0.104)	(0.118)	(0.237)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.075	0.088	0.163
Observations	$4,\!598$	4,598	4,598
Panel B: Foreign Origin Students			
Panel B: Foreign Origin Students Foreign origin teacher	0.167	0.272**	0.181
Panel B: Foreign Origin Students Foreign origin teacher	0.167 (0.138)	0.272^{**} (0.137)	0.181 (0.271)
Panel B: Foreign Origin Students Foreign origin teacher Class FE	0.167 (0.138) Yes	0.272^{**} (0.137) Yes	0.181 (0.271) Yes
Panel B: Foreign Origin Students Foreign origin teacher Class FE Student controls	0.167 (0.138) Yes Yes	0.272** (0.137) Yes Yes	0.181 (0.271) Yes Yes
Panel B: Foreign Origin Students Foreign origin teacher Class FE Student controls Teacher controls	0.167 (0.138) Yes Yes No	0.272** (0.137) Yes Yes Yes	0.181 (0.271) Yes Yes Yes
Panel B: Foreign Origin Students Foreign origin teacher Class FE Student controls Teacher controls Student FE	0.167 (0.138) Yes Yes No No	0.272** (0.137) Yes Yes Yes No	0.181 (0.271) Yes Yes Yes Yes
Panel B: Foreign Origin Students Foreign origin teacher Class FE Student controls Teacher controls Student FE Adjusted R ²	0.167 (0.138) Yes Yes No No 0.091	0.272** (0.137) Yes Yes Yes No 0.098	0.181 (0.271) Yes Yes Yes Yes 0.240

Table 8: Effect of Foreign Origin Teacher on Students' Perceptionthat the Teacher Has Class Under Control

Panel A: Native Students	(1)	(2)	(3)
Foreign origin teacher	-0.043	-0.188	-0.119
	(0.125)	(0.148)	(0.292)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.237	0.250	0.285
Observations	4,735	4,735	4,735
Panel B: Foreign Origin Students			
Foreign origin teacher	0.245^{*}	0.241^{*}	0.259
	(0.135)	(0.141)	(0.314)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.212	0.213	0.311
Observations	2,478	2,478	2,478

Table 7 and 8 display the results on role model effects. Panel B of Table 7 shows that foreign origin students report more often that their teacher tries to understand their point of view, and then tells them what he/she would do if the teacher has a foreign origin. Native students seem to be more likely to feel supported by their native teacher. This differential evaluation persists in a more general question such as "I think my German class teacher has the class under control" where foreign origin students perceive foreign origin teachers more favorably.

Overall, the results in our preferred specification (3) are insignificant, which is partly due to the limited variation in outcome variables.²³ Nevertheless, the pattern displayed supports a role model effect. Foreign origin students are more likely to feel understood by foreign origin than native teachers.²⁴ Therefore, I conclude that the strong positive effect of having a foreign origin teacher on foreign origin students can partly be attributed to a role model effect that motivates foreign origin students differently than their native counterparts.

5.2.3 Language-Specific Skills

To understand why there is an effect of teachers' origin, not only on foreign origin students but also on native student, this section focuses on an obvious characteristic in which native and foreign origin teachers differ: Language skills. Native teachers—by definition—only have German as a mother tongue, while foreign origin teachers often grew up learning more than one language. This difference in exposure to languages could have important implications for their language teaching skills. Around 63 percent of foreign origin teachers in the sample are born in Germany. Going through the German education system requires them to be fluent in German, but the experience of learning German as a second language, or the early exposure to more than one language, might affect the way they understand and

²³The results of Table 7–8 are comparable when controlling for students test scores.

²⁴Other perception variables such as "I think my German teacher encourages me to ask questions" (Table A13), "I think my German teacher expects me to try my very best" (Table A14) or "I think my German teacher is aware of everything that happens in class." display similar results (Table A15).

teach languages.²⁵ The "conscious competence" learning model, for example, describes the psychological progress from incompetence to competence in a skill. The model implies that "conscious competence" teachers (e.g., bilingual teachers who have consciously learned German language skills), are better teachers than "unconscious competence" teachers (e.g., native teachers), because "unconscious competence" teachers can have difficulties in explaining the skill that has become largely instinctual to them (Robinson, 1974). To elicit the effect of such a potential language specific skill, I test if foreign origin teachers who report a mother tongue other than German affect students' reading comprehension differently.

Panel A: All Students	(1)	(2)	(3)
Bilingual teacher	0.291**	** 0.333**	* 0.431**
0	(0.073)	(0.098)	(0.169)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.377	0.378	0.661
Observations	$7,\!351$	$7,\!351$	$7,\!351$
Panel B: Native Students			
Bilingual teacher	0.352**	** 0.467**	* 0.533**
0	(0.108)	(0.127)	(0.263)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.362	0.363	0.649
Observations	4,812	4,812	$4,\!812$
Panel C: Foreign Origin Students			
Bilingual teacher	0.239**	° 0.196	0.516**
	(0.117)	(0.133)	(0.211)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R ²	0.394	0.394	0.656
Observations	2,539	2,539	2,539

Table 9: Effect of Bilingual Teachers on
Reading Comprehension

Notes: – Results are obtained from OLS regressions. All reported standard errors are clustered at the teacher level. – *** p<0.01, ** p<0.05, * p<0.10.

Table 9 displays the effect of having a bilingual teacher, i.e., a teacher with a mother

 $^{^{25}}$ While the economic literature on bilingualism is scarce, a study by Clifton-Sprigg (2015) finds no differences in children's cognitive and non-cognitive skills based on the number of languages spoken at home.

tongue other than German, on reading comprehension scores. In comparison to the baseline regression in Table 4, Panel A, the effect of having a bilingual teacher is larger and more significant than the effect of having a foreign origin teacher. Further, the table reveals that the effect is particularly large for native students. Their reading comprehension increases by 0.53, i.e., 0.42 standard deviations, when they have a bilingual teacher. Nevertheless, the positive effect holds for both native and foreign origin students.

As a falsification test, I test if having a bilingual Math teacher influences math test scores in a similar way (Table A16). If bilingual teachers have language-specific skills, rather than being a particularly positive selection of teachers, the effect on language comprehension should be larger than on analytical math skills. The regressions on math test scores reveal no effect of having a bilingual teacher once teacher characteristics are included.²⁶ The effect of bilingual teachers is smaller and indistinguishable from zero for both sub-samples. This finding supports the notion that the effect of bilingual teachers is language-specific rather than driven by a particular positive selection of bilingual teachers.

In summary, the positive effect is specific to reading comprehension and particularly strong for bilingual teachers. While most studies discuss matching effects between students and teachers, these results suggest that foreign origin teachers who grew up bilingually are especially equipped to teach languages. Notably, this language-specific skill is beneficial for both native and foreign origin students.

5.3 Alternative Explanations

In this section, I report results from alternative specifications to provide additional robustness checks and explore the heterogeneity of the findings.

First, I investigate if a particular group of foreign origin teachers is driving the effect. I test if first generation immigrant teachers and second and higher generation immigrant teachers affect reading performance differently. A strong effect of foreign born teachers could hint at a positive selection, e.g., because of better teacher training abroad. Table 10

²⁶The samples on reading comprehension and mathematics scores do not match perfectly as the NEPS data contain more observations with information on Math teacher than German teacher. More information on Math teachers' characteristics are displayed in Tables A5 and A6.

shows that better educated foreign born teachers cannot explain the findings. The effect of first generation immigrant teachers is larger for native students but overall the effects for both teacher groups are comparable in size and insignificant. Alternatively, the effect could be driven by a particular well equipped immigrant group. Testing for differential effects across language groups, I find positive and large point estimates for all language groups (Table A17). Accordingly, the results cannot be explained by one particularly motivated or able immigrant group.

Reading Co	mprehe	ension	
Panel A: All Students	(1)	(2)	(3)
Immigrant background (Ref.: None)			
First generation	0.169^{*}	0.393^{**}	* 0.257
	(0.100)	(0.129)	(0.215)
Second and higher generation	0.178^{*}	0.147^{*}	0.260
	(0.092)	(0.083)	(0.162)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.377	0.378	0.660
Observations	$7,\!351$	$7,\!351$	$7,\!351$
Panel B: Native Students			
Immigrant background (Ref.: None)			
First generation	0.317^{**}	* 0.544**	* 0.313
	(0.114)	(0.154)	(0.236)
Second and higher generation	0.229	0.171	0.264
	(0.147)	(0.136)	(0.278)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.362	0.363	0.647
Observations	4,812	4,812	$4,\!812$
Panel C: Foreign Origin Students			
Immigrant background (Ref.: None)			
First generation	0.056	0.227	0.497
	(0.159)	(0.182)	(0.355)
Second and higher generation	0.144	0.093	0.327
	(0.112)	(0.104)	(0.224)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.394	0.393	0.655
Observations	2,539	2,539	2,539

Table 10: Effect of Immigrant Generation on
Reading Comprehension

Notes: – Results are obtained from OLS regressions. All reported standard errors are clustered at the teacher level. – *** p<0.01, ** p<0.05, * p<0.10.

Another potential concern is that the results are caused by a different selection of foreign origin teachers in comparison to native teachers. While it is beyond the scope of this paper to discuss potential differences in unobservables, I can proxy the intrinsic motivation by controlling for variables that capture aspects of the teacher's career choice and cultural orientation (Table A18–A19).²⁷ Different motivations for becoming a teacher do not influence the students' performance significantly while teachers who belief in the relevance of multicultural ideals have a positive effect on native students reading comprehension. The general pattern of the foreign origin teacher effect remains in both specifications.

	0	-	
Panel A: All Students	(1)	(2)	(3)
Ever had a foreign origin teacher	0.097	0.123^{*}	0.243*
	(0.060)	(0.063)	(0.138)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R ²	0.377	0.377	0.660
Observations	$7,\!351$	7,351	$7,\!351$
Panel B: Native Students			
Ever had a foreign origin teacher	0.103	0.121	0.208
	(0.081)	(0.091)	(0.220)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.361	0.362	0.647
Observations	4,812	4,812	4,812
Panel C: Foreign Origin Students			
Ever had a foreign origin teacher	0.126	0.140	0.422**
	(0.102)	(0.105)	(0.210)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.394	0.394	0.655
Observations	2,539	2,539	2,539

Table 11: Effect of Ever Being Taught By A Foreign OriginTeacher on Reading Comprehension

Notes: – Results are obtained from OLS regressions. All reported standard errors are clustered at the teacher level. – *** p < 0.01, ** p < 0.05, * p < 0.10.

The results are also robust to the inclusion of the students' math test scores (see Table A21). Math test scores proxy student-year-specific performance and accordingly provide a lower bound estimate of the foreign origin teacher effect if there are positive spillover effects, i.e., if better reading comprehension increases math test scores. Further, reading

²⁷As native and foreign origin differ significantly with respect to grades at the first state examination, I test if foreign origin teachers who receive better grades are driving the effect. Table A20 displays the results and hints at an effect that is rather driven by poorly performing foreign origin teachers than foreign origin teachers who perform particularly well.

comprehension is a cumulative skill and a teacher's positive effect should also pay off in the following school years. Therefore, Table 11 illustrates the effect of ever having had a foreign origin teacher. It illustrates that the positive foreign origin teacher effect persists if one considers the subsequent reading comprehension scores. The long term effect of foreign origin teachers is particularly strong for foreign origin students suggesting a long lasting role model effect.

Panel A: All Students	(1)	(2)	(3)
Foreign origin teacher	0.106	0.125	0.191
0 0	(0.109)	(0.088)	(0.192)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.151	0.151	0.436
Observations	$3,\!674$	$3,\!674$	$3,\!674$
Panel B: Native Students			
Foreign origin teacher	0.146	0.037	-0.103
0 0	(0.092)	(0.161)	(0.448)
Class FE	No	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.058	0.133	0.409
Observations	2,406	2,406	2,406
Panel C: Foreign Origin Students			
Foreign origin teacher	-0.011	0.115	0.661**
	(0.089)	(0.094)	(0.212)
Class FE	No	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.065	0.124	0.442
Observations	1,268	1,268	1,268

Table 12: Effect of Foreign Origin Teacher onReading Comprehension of Best Students

Notes: – Results are obtained from OLS regressions. All reported standard errors are clustered at the teacher level. – *** p < 0.01, ** p < 0.05, * p < 0.10.

I further test if there are heterogeneous effects of having a foreign origin teacher with respect to students' proficiency. More specifically, I divide the samples by the median reading comprehension score. Table 12 reports the results for the 50 percent best students of each sample and Table 13 summarizes the results for the worst students. In Table 12, the overall positive effect of foreign origin teachers is small. For above median native students the effect is even negative but insignificant. For above median foreign origin

Panel A: All Students	(1)	(2)	(3)
Foreign origin teacher	0.118^{*}	0.163^{**}	0.310
0 0	(0.069)	(0.079)	(0.194)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.228	0.228	0.340
Observations	$3,\!677$	$3,\!677$	$3,\!677$
Panel B: Native Students			
Foreign origin teacher	0.107	0.323**	* 0.503**
0 0	(0.077)	(0.118)	(0.251)
Class FE	No	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.041	0.192	0.298
Observations	2,406	2,406	$2,\!406$
Panel C: Foreign Origin Students			
Foreign origin teacher	-0.191^{*}	-0.113	0.045
	(0.102)	(0.124)	(0.299)
Class FE	No	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.056	0.145	0.271
Observations	1,271	1,271	1,271

Table 13: Effect of Foreign Origin Teacher onReading Comprehension of Worst Students

students, however, the effect displayed is large and significant. Table 13 illustrates the results for the 50 percent of students with the worst reading performance and shows that the effects are reversed for this sample. Here, the worst native students have higher reading comprehension scores if their teacher has a foreign origin, while the effect on the below median foreign origin students is not different from zero.²⁸ This finding adds new evidence to the matching literature which has previously shown that low-performing students benefit particularly from being assigned to race-congruent teachers (Egalite *et al.*, 2015).

 $^{^{28}}$ This finding is confirmed when distinguishing between the academic track (*Gymnasium*) and the other non-academic tracks (*Realschule* and *Hauptschule*) (Table A22 and A23).

6 Conclusion

This paper is the first to provide evidence on a positive effect of having a foreign origin teacher on students' academic achievement. Given the under-performance of an increasing share of foreign origin students, investigating the effect of foreign origin teachers fills an important gap in the economics literature. Using data from the German National Educational Panel Study and exploiting variation in teacher assignment within student, I analyze the effect of having a foreign origin German language teacher on reading comprehension in lower secondary school. In doing so, I show that objectively measured reading test scores are positively affected by having a foreign origin teacher.

For foreign origin students, I find evidence for a role model effect, in line with studies on race and gender matching (e.g., Dee, 2004; Bettinger and Long, 2005; Paredes, 2014). Foreign origin students improve their reading comprehension significantly and perceive their teacher more favorably if the teacher is of foreign origin. Adding to the literature on matching effects, this study shows that role model effects can exist for "imperfect" matches, meaning that students' and teachers' countries of origin do not have to be identical for role model effects to emerge. Overall, the improvement in foreign origin students' language fluency can have important implications for lifelong outcomes such as productivity and wages as language proficiency facilitates the adaptation of skills in the job market.

Most notably, the study finds that bilingual foreign origin teachers increase students' reading comprehension scores universally. Ruling out alternative explanations, I argue that bilingual teachers have language-specific skills that make them particularly well-equipped to teach languages both to native and foreign origin students. Therefore, the study contributes to the literature by establishing a role for bilingual teachers in the context of language acquisition. It should encourage researchers to study the advantages of bilingualism in the education system more rigorously as the knowledge and experience of bilingual teachers can be used to develop language teaching styles that improve the reading comprehension of both native and foreign origin students.

References

- AARONSON, D., BARROW, L. and SANDER, W. (2007). Teachers and Student Achievement in the Chicago Public High Schools. *Journal of Labor Economics*, **25** (1), 95–135.
- ALESINA, A., CARLANA, M., LA FERRARA, E. and PINOTTI, P. (2018). Revealing stereotypes: Evidence from immigrants in schools. *NBER Working Paper 25333*.
- ALGAN, Y., DUSTMANN, C., GLITZ, A. and MANNING, A. (2010). The economic situation of first and second–generation immigrants in France, Germany and the United Kingdom. *The Economic Journal*, **120** (542), 4–30.
- ANGRIST, J. D. and KRUEGER, A. B. (1991). Does compulsory school attendance affect schooling and earnings? *The Quarterly Journal of Economics*, **106** (4), 979–1014.
- ANTECOL, H., EREN, O. and OZBEKLIK, S. (2015). The Effect of Teacher Gender on Student Achievement in Primary School. *Journal of Labor Economics*, **33** (1), 63–89.
- ASANO, A. (2008). Are Native and Non-Native English Speaking Tutors Equally Effective? The Australian Economic Review, **41** (1), 56–66.
- BETTINGER, E. P. and LONG, B. T. (2005). Do Faculty Serve as Role Models? The Impact of Instructor Gender on Female Students. *The American Economic Review*, 2 (95), 152–157.
- BLOSSFELD, H.-P. (ed.) (2011). Education as a lifelong process: The German National Educational Panel Study (NEPS), Zeitschrift für Erziehungswissenschaft Sonderheft, vol. 14. Wiesbaden: VS-Verlag.
- BORJAS, G. J. (2000). Foreign-Born Teaching Assistants and the Academic Performance of Undergraduates. *The American Economic Review*, **90** (2), 355–359.
- BUNDESAMT FÜR MIGRATION UND FLÜCHTLINGE (ed.) (2010). Bundesweites Integrationsprogramm: Angebote der Integrationsförderung in Deutschland – Empfehlungen zu ihrer Weiterentwicklung. Paderborn: Bonifatius.
- CANES, B. J. and ROSEN, H. S. (1995). Following in her footsteps? Faculty gender composition and women's choices of college majors. *Industrial and Labor Relations Review*, 48 (3), 486–504.
- CHETTY, R., FRIEDMAN, J. N. and ROCKOFF, J. E. (2014a). Measuring the impacts of teachers I: Evaluating bias in teacher value-added estimates. *American Economic Review*, **104** (9), 2593–2632.
- —, and (2014b). Measuring the Impacts of Teachers II: Teacher Value-Added and Student Outcomes in Adulthood. *The American Economic Review*, pp. 2633–2679.
- CHIN, A., DAYSAL, N. M. and IMBERMAN, S. A. (2013). Impact of bilingual education programs on limited English proficient students and their peers: Regression discontinuity evidence from Texas. *Journal of Public Economics*, **107**, 63–78.
- CHISWICK, B. R. and MILLER, P. W. (1995). The Endogeneity between Language and Earnings: International Analyses. *Journal of Labor Economics*, **13** (2), 246–288.

- CLIFTON-SPRIGG, J. (2015). Best of both worlds? Early cognitive and non-cognitive development of bilingual children. *Unpublished Manuscript*.
- DEE, T. S. (2004). Teachers, Race, and Student Achievement in a Randomized Experiment. *Review of Economics and Statistics*, **86** (1), 195–210.
- (2005). A Teacher like Me: Does Race, Ethnicity, or Gender Matter? The American Economic Review, 2 (95), 158–165.
- (2007). Teachers and the Gender Gaps in Student Achievement. The Journal of Human Resources, 42 (3), 528–554.
- DUSTMANN, C. and FABBRI, F. (2003). Language Proficiency and Labour Market Performance of Immigrants in the UK. *The Economic Journal*, **113** (489), 695–717.
- --, FRATTINI, T. and LANZARA, G. (2012). Educational achievement of second-generation immigrants: An international comparison. *Economic Policy*, **27** (69), 143–185.
- —, MACHIN, S. and SCHÖNBERG, U. (2010). Ethnicity and Educational Achievement in Compulsory Schooling. *The Economic Journal*, **120** (546), 272–297.
- and VAN SOEST, A. (2001). Language Fluency and Earnings: Estimation with Misclassified Language Indicators. *The Review of Economics and Statistics*, **83** (4), 663–674.
- EGALITE, A. J., KISIDA, B. and WINTERS, M. A. (2015). Representation in the classroom: The effect of own-race teachers on student achievement. *Economics of Education Review*, **45**, 44–52.
- FAIRLIE, R. W., HOFFMANN, F. and OREOPOULOS, P. (2014). A Community College Instructor Like Me: Race and Ethnicity Interactions in the Classroom. American Economic Review, 104 (8), 2567–2591.
- FLEISHER, B., HASHIMOTO, M. and WEINBERG, B. A. (2002). Foreign GTAs Can Be Effective Teachers of Economics. *The Journal of Economic Education*, **33** (4), 299–325.
- FURUYA, Y., NOORADDINI, M. I. and WANG, WENJING: WASLIN, MICHELE (2019). A Portrait of Foreign-Born Teachers in the United States. *George Mason University*, *Institute for Immigration Research*.
- GEAY, C., MCNALLY, S. and TELHAJ, S. (2013). Non-native Speakers of English in the Classroom: What Are the Effects on Pupil Performance? *The Economic Journal*, 123 (570), F281–F307.
- GEHRER, K., ZIMMERMANN, S., ARTELT, C. and WEINERT, S. (2012). The Assessment of Reading Competence (Including Sample Items for Grade 5 and 9): Scientific Use File 2012, Version 1.0.0. Bamberg: University of Bamberg, National Educational Panel Study.
- GERSHENSON, S., HOLT, S. B. and PAPAGEORGE, N. W. (2016). Who believes in me? The effect of student-teacher demographic match on teacher expectations. *Economics* of *Education Review*, **52**, 209–224.

- HAYFRON, J. E. (2001). Language training, language proficiency and earnings of immigrants in Norway. *Applied Economics*, **33** (15), 1971–1979.
- HINNERICH, B. T., HÖGLIN, E. and JOHANNESSON, M. (2015). Discrimination against students with foreign backgrounds: Evidence from grading in Swedish public high schools. *Education Economics*, **23** (6), 660–676.
- HOFFMANN, F. and OREOPOULOS, P. (2009). A Professor like Me: The Influence of Instructor Gender on College Achievement. *The Journal of Human Resources*, **44** (2), 479–494.
- ISPHORDING, I. E. (2014). Disadvantages of linguistic origin Evidence from immigrant literacy scores. *Economics Letters*, **123** (2), 236–239.
- KISS, D. (2013). Are immigrants and girls graded worse? Results of a matching approach. *Education Economics*, **21** (5), 447–463.
- MALECKI, A., SCHNEIDER, CHRISTOPH, VOGEL, SILVIA and WOLTERS, M. (2014). Schulen auf einen Blick: Ausgabe 2014. Wiesbaden: Statistisches Bundesamt.
- NOREISCH, K. (2006). School catchment area evasion: The case of Berlin, Germany. Journal of Education Policy, 22 (1), 69–90.
- OECD (ed.) (2018). The Resilience of Students with an Immigrant Background: Factors that Shape Well-Being. Paris: OECD Publishing.
- PAREDES, V. (2014). A teacher like me or a student like me? Role model versus teacher bias effect. *Economics of Education Review*, **39**, 38–49.
- RIVKIN, S. G., HANUSHEK, E. A. and KAIN, J. F. (2005). Teachers, Schools, and Academic Achievement. *Econometrica*, **73** (2), 417–458.
- ROBINSON, W. L. (1974). Conscious competency The mark of a competent instructor. *Personnel Journal*, **53**, 538–539.
- ROCKOFF, J. E. (2004). The impact of individual teachers on student achievement: Evidence from panel data. *American Economic Review*, **94** (2), 247–252.
- ROTHSTEIN, J. (2010). Teacher quality in educational production: Tracking, decay, and student achievement. *The Quarterly Journal of Economics*, **125** (1), 175–214.
- SCHNEPF, S. V. (2007). Immigrants' educational disadvantage: An examination across ten countries and three surveys. *Journal of Population Economics*, **20** (3), 527–545.
- SEAH, K. K. C. (2018a). Do You Speak My Language? The Effect of Sharing a Teacher's Native Language on Student Achievement. *IZA Discussion Paper No. 11685*.
- (2018b). Immigrant educators and students' academic achievement. Labour Economics, 51, 152–169.
- SPRIETSMA, M. (2013). Discrimination in grading: Experimental evidence from primary school teachers. *Empirical Economics*, **45** (1), 523–538.

- STATISTISCHES BUNDESAMT (2012). Bevölkerung und Erwerbstätigkeit: Bevölkerung mit Migrationshintergrund - Ergebnisse des Mikrozensus 2012, Fachserie 1, vol. 2.2.
- STATISTISCHES BUNDESAMT (2018). Sonderauswertung. Bevölkerung und Erwerbstätigkeit: Bevölkerung mit Migrationshintergrund - Ergebnisse des Mikrozensus 2012-2016. Wiesbaden.
- STEINHAUER, H. W. and ZINN, S. (2016). NEPS Technical report for weighting: Weighting the sample of starting cohort 3 of the National Educational Panel Study (Wave 1 to 3). *NEPS Working Paper No. 63.*

Appendix



Figure A1: German School System

Notes: - Own illustration.

	(1) Female student		(3) Class repeated	(4) Math testscore
Foreign origin student-teacher interaction	-0.007 (0.055)	-2.995 (4.602)	-0.008 (0.015)	-0.007 (0.107)
Observations	7,313	6,969	7,351	7,346

Table A1: Student-Teacher Sorting by Foreign Origin

Notes: – Each estimate is from a regression of the corresponding student characteristic on a dummy indicating the teacher's origin, a dummy indicating the student's origin and the interaction between these two variables. Robust standard errors are in parentheses. – *** p < 0.01, ** p < 0.05, * p < 0.10.

	(1) Female student		(3) Class repeated	(4) Math testscore
Foreign origin teacher	$0.014 \\ (0.047)$	-1.193 (1.062)	-0.010 (0.012)	0.166^{*} (0.088)
Observations	7,313	6,969	7,351	7,346

Table A2: Sorting by Foreign Origin of Teacher

Notes: – Each estimate is from a regression of the corresponding student characteristic on a dummy indicating the teacher's origin in a classroom. Robust standard errors are in parentheses. – *** p < 0.01, ** p < 0.05, * p < 0.10.

	Native	Foreign Origin	Difference	Std. Error
Main outcome variable				
Reading comprehension score	0.421	0.064	0.357	$(0.031)^{***}$
Student characteristics				
Female	0.492	0.507	-0.015	(0.012)
Grade	6.786	6.696	0.090	$(0.039)^{**}$
Birth month	6.520	6.584	-0.065	(0.084)
Birth year	1999.538	1999.437	0.101	$(0.015)^{***}$
Grade repeated	0.020	0.028	-0.008	$(0.004)^{**}$
Household size	4.392	4.505	-0.112	$(0.033)^{***}$
Other outcome variables				
Mathematical test score	0.470	0.041	0.429	$(0.029)^{***}$
German grade	4.456	4.260	0.195	$(0.025)^{***}$
German teacher tries to understand my point	3.271	3.281	-0.010	(0.027)
German teacher has class under control	3.674	3.712	-0.038	(0.029)
Native student obs.	4,812			
Foreign origin student obs.	2,539			

Table A3: Student Characteristics, t-test

Notes: – The table shows descriptive statistics for the two sub-samples of native and foreign origin students, as well as the difference in mean values between the two samples. Significance stars indicate the result of the respective t-test. – *** p < 0.01, ** p < 0.05, * p < 0.10.

Table A4: German Language Teacher Characteristics, t-test

	Native	Foreign Origin	Difference	Std. Error
Female	0.781	0.847	-0.066	(0.021)***
Birth year	1967.650	1975.141	-7.491	$(0.596)^{***}$
Age when profession was chosen	1.574	1.119	0.455	$(0.058)^{***}$
Grade at first state exam	-0.006	-0.172	0.166	(0.046)***
German language studies	1.052	0.993	0.059	(0.023)***
Language studies	0.485	0.262	0.223	(0.036)***
State of higher education entrance qualification	5.964	5.772	0.192	(0.168)
School type	4.579	5.171	-0.592	$(0.077)^{***}$
Native teacher obs.	6,947			
Foreign origin teacher obs.	404			

Notes: – The table shows descriptive statistics for the two sub-samples of native and foreign origin German language teachers, as well as the difference in mean values between the two samples. Significance stars indicate the result of the respective t-test. – *** p < 0.01, ** p < 0.05, * p < 0.10.

	Mean	Std. Dev.	Min	Max
Foreign origin	0.057	0.23	0	1
Second and higher generation immigrant	0.033	0.18	0	1
First generation immigrant	0.024	0.15	0	1
Bilingual	0.020	0.14	0	1
Slavic	0.006	0.08	0	1
Romance	0.005	0.07	0	1
Others	0.009	0.09	0	1
Female	0.577	0.49	0	1
Age	44.712	12.04	22	70
Birth decade				
1940s	0.031	0.17	0	1
1950s	0.320	0.47	0	1
1960s	0.202	0.40	0	1
1970s	0.198	0.40	0	1
1980s	0.249	0.43	0	1
Age when profession was chosen				
Between 0-14	0.152	0.36	0	1
Between 15-19	0.450	0.50	0	1
Between 20-24	0.180	0.38	0	1
After 25	0.097	0.30	0	1
Grade at first state exam	-0.035	0.85	-2	3
Teacher-year observations	8,937			
Teacher observations	818			

Table A5: Descriptive Statistics – Math Teachers

Notes: – The descriptive statistics are weighted to account for differences in the number of observations per teacher.

	Native	Foreign Origin	Difference	Std. Error
Female	0.556	0.399	0.158	$(0.025)^{***}$
Birth year	1965.982	1974.273	-8.291	(0.588)***
Age when profession was chosen	1.514	1.232	0.282	(0.058)***
Grade at first state exam	-0.054	0.030	-0.084	$(0.042)^{**}$
State of higher education entrance qualification	6.088	5.754	0.335	$(0.166)^{**}$
School type	4.686	4.507	0.179	$(0.075)^{**}$
Native teacher obs.	8,523			
Foreign origin teacher obs.	414			

Table A6: Math Teacher Characteristics, t-test

Notes: – The table shows descriptive statistics for the two sub-samples of native and foreign origin Math teachers, as well as the difference in mean values between the two samples. Significance stars indicate the result of the respective t-test. – *** p < 0.01, ** p < 0.05, * p < 0.10.

	(1)	(2)	(3)
Foreign origin teacher	0.175^{**}	0.232^{***}	0.259^{*}
Student characteristics	0.114***	0.116***	(0.137)
Foreign origin	(0.029) 0.182***	(0.029)	
	(0.032)	(0.032)	—
February	-0.057	-0.056	-
March	$(0.066) \\ 0.013$	$(0.066) \\ 0.014$	_
April	(0.064) -0.044	(0.064) -0.044	_
Mari	(0.067)	(0.067)	
-	(0.070)	(0.071)	_
June	-0.024 (0.066)	(0.026)	_
July	0.112^{*} (0.065)	0.112^{*} (0.065)	-
August	(0.109)	0.112^{*}	-
September	0.092	0.097	-
October	(0.072) 0.278^{***}	(0.072) 0.275^{***}	-
November	$(0.075) \\ 0.144^*$	$(0.075) \\ 0.141^*$	_
December	(0.075) 0.155**	(0.075) 0.157**	_
	(0.074)	(0.074)	
1995	0.054	0.082	-
1997	$(0.401) \\ -0.124$	(0.426) -0.101	_
1998	$(0.345) \\ -0.197^{***}$	(0.348) -0.193***	_
2000	(0.054)	(0.054)	
2000	(0.037)	(0.037)	—
2001	0.325^{-} (0.168)	0.324^{*} (0.169)	-
2002	1.664^{***} (0.059)	1.667^{***} (0.059)	—
Grade (Ref.: 5)	0.611***	0.574***	0 501***
	(0.039)	(0.040)	(0.068)
9	(0.040)	(0.042)	(0.069)
Grade repeated	-0.081 (0.092)	-0.083 (0.092)	-0.209 (0.171)
Household size	-0.029^{***}	-0.029^{***}	-0.067^{**} (0.032)
Teacher characteristics	(01010)	0.069***	0.065*
Grade at first state exam	_	(0.023)	(0.037)
Female	_	-0.017 (0.044)	-0.079 (0.061)
Age	—	0.005 (0.008)	-0.003 (0.013)
Birth decade (Ref.: 1940s)	_	0.103	0.130
19908		(0.100)	(0.176)
1960s	-	(0.139) (0.146)	(0.250)
1970s	-	-0.167 (0.209)	0.029 (0.336)
1980s	-	-0.036	0.175
Age job was chosen (Ref.: Before 10) Between 15-19	_	0.022	0.039
Between 20-24	_	(0.061) 0.074	$(0.106) \\ 0.131$
After 25	_	$(0.068) \\ -0.090$	$(0.122) \\ 0.057$
German studies	_	(0.081) -0.030	(0.126) -0.009
Other language studies	_	(0.097) 0.103**	(0.108)
Class EE	37	(0.051)	(0.083)
Glass FE Student FE	Yes No	Yes No	Yes Yes
Adjusted R ² Observations	$0.377 \\ 7,351$	0.378 7,351	$0.660 \\ 7,351$

Table A7: Effect of Foreign Origin Teacher onReading Comprehension – All Students

	Nati	ive Students		Foreign Origin Stu		idents
	(1)	(2)	(3)	(4)	(5)	(6)
Foreign origin teacher	0.260^{**}	* 0.315 **	(0.279)	0.115 (0.087)	0.129	0.375^{*}
Student characteristics	(0.100)**	(0.101)	(0.220)	(0.007)	(0.000)	(0.100)
Female	(0.128) (0.037)	(0.129) (0.037)	. –	(0.087) (0.054)	(0.090) (0.054)	-
Birth month (Ref.: January) February	-0.128	-0.126	_	0.151	0.145	_
March	$(0.084) \\ -0.065$	(0.085) -0.066	_	$(0.126) \\ 0.225^*$	$(0.127) \\ 0.235^*$	_
April	(0.084)	(0.085)	_	(0.126) 0.206	(0.127) 0.213	_
	(0.085)	(0.086)		(0.160)	(0.161)	
Мау	(0.087)	(0.083)	_	(0.152)	(0.153)	_
June	-0.069 (0.087)	-0.067 (0.087)	-	0.078 (0.125)	0.079 (0.126)	-
July	0.111	(0.091)	-	0.227^{*}	0.224^{*}	-
August	0.040	0.045	-	0.384**	** 0.396**	* -
September	(0.087) 0.021	(0.088) 0.030	_	(0.137) 0.364^{**}	(0.138) * 0.372**	_
October	(0.091) 0.257^{**}	(0.091) ** 0.262**	* _	(0.147) 0.372^{**}	(0.147) ** 0.363**	_
November	(0.097)	(0.097)	_	(0.144)	(0.145)	_
	(0.097)	(0.098)		(0.149)	(0.151)	
December	$\begin{array}{c} 0.122 \\ (0.097) \end{array}$	(0.126) (0.098)	_	0.294° (0.153)	(0.303^{**}) (0.154)	-
Birth year (Ref.: 1999) 1995	0.154	0.286	_	_	_	_
1007	(0.292)	(0.326)	*	0.245	0.227	
1997	(0.083)	(0.932)	_	(0.343)	(0.342)	_
1998	-0.162^{**} (0.081)	(0.081)	`_	-0.226^{**} (0.086)	(0.086)	_
2000	0.165^{**} (0.052)	(0.052)	* _	0.188^{**} (0.070)	(0.071)	* _
2001	0.343	0.347	-	0.577^{*}	0.620^{*}	-
2002	(0.217) 1.503^{**}	** 1.502**	** _	(0.340)	(0.342)	-
Grade (Ref.: 5)	(0.089)	(0.089)				
7	0.571^{**} (0.048)	(0.544^{**})	(0.074)	* 0.642** (0.067)	(0.597^{**})	* 0.610*** (0.136)
9	-0.153^{**}	**-0.123**	(0.082)	0.042	0.005	-0.093
Grade repeated	(0.049) -0.055	-0.069	-0.288	-0.151	-0.131	0.009
Household size	$(0.137) - 0.025^*$	$(0.136) - 0.026^*$	(0.289) - 0.035	(0.153) -0.055^{**}	(0.156) ** -0.054 **	(0.275) *-0.109**
Teacher characteristics	(0.014)	(0.014)	(0.042)	(0.018)	(0.018)	(0.053)
Grade at first state exam	—	0.048^{*}	0.079^{*}	—	0.056^{*}	0.067
Female	-	-0.054	-0.111	-	0.142*	0.011
Age	_	(0.049) 0.006	(0.074) 0.000	_	(0.086) 0.002	(0.119) 0.001
Birth decade (Ref.: 1940s)		(0.010)	(0.017)		(0.012)	(0.022)
1950s	-	0.140 (0.118)	0.102 (0.207)	_	0.040	0.012 (0.328)
1960s	-	-0.042	-0.059	-	-0.414^{*}	-0.178
1970s	-	(0.181) -0.155	(0.321) -0.105	_	(0.234) -0.182	(0.414) -0.093
1980s	_	(0.251) 0.011	$(0.424) \\ 0.023$	_	(0.328) -0.176	$(0.588) \\ 0.045$
Age job was chosen (Ref · Before 10)		(0.326)	(0.556)		(0.428)	(0.759)
Between 15-19	-	0.031	0.040	-	0.085	0.056
Between 20-24	-	(0.074) 0.131	0.129)	_	0.087	(0.162) 0.160
After 25	_	(0.084) -0.160	$(0.147) \\ 0.033$	_	$(0.103) \\ 0.142$	(0.179) 0.153
German studies	_	(0.102) -0.191	$(0.156) \\ 0.003$	_	(0.123) 0.104	$(0.196) \\ -0.144$
Other language studies	_	(0.138) 0.116*	(0.165)	_	(0.113)	(0.181)
Chine Language studies	-	(0.070)	(0.032)	-	(0.074)	(0.134)
Class FE Student FE	Yes No	Yes No	Yes Yes	Yes No	Yes No	Yes Yes
Adjusted \mathbb{R}^2 Observations	$0.362 \\ 4.812$	$0.363 \\ 4.812$	0.648	0.394	$0.394 \\ 2.539$	0.655
	-,	-,02	-,	2,000	2,000	_,000

Table A8: Effect of Foreign Origin Teacher on Reading Comprehension – By Student Origin

Notes: – Results are obtained from OLS regressions. All reported standard errors are clustered at the teacher level. – *** p < 0.01, ** p < 0.05, * p < 0.10.

_

	(1)	(0)	(2)
Panel A: All Students	(1)	(2)	(3)
Foreign origin teacher	-0.031	-0.028	-0.033
	(0.064)	(0.063)	(0.086)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.469	0.469	0.737
Observations	8,937	8,937	8,937
Panel B: Native Students			
Foreign origin teacher	-0.089	-0.123	-0.094
	(0.078)	(0.078)	(0.093)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.438	0.438	0.722
Observations	5,952	5,952	5,952
Panel C: Foreign Origin Students			
Foreign origin teacher	0.065	0.050	-0.018
	(0.114)	(0.117)	(0.207)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.502	0.501	0.746
Observations	2,985	2,985	2,985

Table A9: Effect of Foreign OriginTeacher on Mathematics

Panel A: All Students	(1)	(2)	(3)
Linguistic Distance	0.002^{*}	0.002^{*}	0.003
	(0.001)	(0.001)	(0.003)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.385	0.386	0.657
Observations	$6,\!498$	6,498	$6,\!498$
Panel B: Native Students			
Linguistic Distance	0.004***	0.005^{***}	0.005
	(0.001)	(0.002)	(0.003)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.365	0.366	0.649
Observations	4,709	4,709	4,709
Panel C: Foreign Origin Students			
Linguistic Distance	-0.001	-0.001	-0.018
	(0.002)	(0.003)	(0.014)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.414	0.413	0.627
Observations	1,789	1,789	1,789

Table A10: Effect of Linguistic Distance between Teacherand Student on Reading Comprehension

Panel A: Native Students	(1)	(2)	(3)
Language match	0.226**	* 0.261**	*-0.103
	(0.060)	(0.096)	(0.340)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.229	0.225	0.653
Observations	$3,\!101$	3,101	3,101
Panel B: Foreign Origin Students			
Language match	-0.238	-0.245	0.032
	(0.483)	(0.532)	(0.333)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.152	0.148	0.593
Observations	$1,\!606$	$1,\!606$	$1,\!606$

Table A11: Effect of Language Match between Teacher and
Student on German Grade (Harmonized Sample)

Notes: – Results are obtained from OLS regressions. The sample of the analysis is restricted to students taking the reading competence test (Table 4). All reported standard errors are clustered at the teacher level. – *** p<0.01, ** p<0.05, * p<0.10.

Table	A12:	Effect	of I	Foreign	Origin
]	Teacher	on	Germa	n Grade

Panel A: Native Students	(1)	(2)	(3)
Foreign origin teacher	-0.014	-0.056	-0.080
	(0.077)	(0.051)	(0.081)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.224	0.223	0.640
Observations	5,757	5,757	5,757
Panel B: Foreign Origin Students			
Foreign origin teacher	-0.173^{*}	-0.180^{*}	-0.111
	(0.094)	(0.101)	(0.129)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.174	0.175	0.625
Observations	2,815	2,815	2,815

Panel A: Native Students	(1)	(2)	(3)
Foreign origin teacher	-0.293^{**}	-0.276^{**}	-0.158
	(0.128)	(0.131)	(0.247)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.083	0.093	0.201
Observations	4,622	$4,\!622$	$4,\!622$
Panel B: Foreign Origin Students			
Foreign origin teacher	0.140	0.240	0.010
	(0.153)	(0.149)	(0.308)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.062	0.067	0.221
Observations	$2,\!435$	2,435	2,435

Table A13: Effect of Foreign Origin Teacher on Students' Perceptionthat the Teacher Encourages Them

Table A14: Effect of Foreign Origin Teacher on Students' Perceptionthat the Teacher Expects Them To Try Their Best

Panel A: Native Students	(1)	(2)	(3)
Foreign origin teacher	0.045	0.020	-0.009
	(0.096)	(0.091)	(0.196)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.072	0.071	0.174
Observations	4,724	4,724	4,724
Panel B: Foreign Origin Students			
Foreign origin teacher	0.043	0.043	0.292
	(0.097)	(0.115)	(0.256)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.075	0.071	0.190
Observations	2,479	$2,\!479$	$2,\!479$

Panel A: Native Students	(1)	(2)	(3)
Foreign origin teacher	-0.215	-0.274^{*}	-0.356
	(0.135)	(0.142)	(0.269)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R ²	0.170	0.180	0.265
Observations	4,754	4,754	4,754
Panel B: Foreign Origin Students			
Foreign origin teacher	-0.042	0.028	0.126
	(0.119)	(0.133)	(0.289)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R ²	0.138	0.144	0.272
Observations	2.493	2.493	2.493

 Table A15: Effect of Foreign Origin Teacher on Students'

 Perception that the Teacher Notices Everything

	Table A16:	Effect of	Bilingual	Teacher	on	Mathematics
--	------------	-----------	-----------	---------	----	-------------

Panel A: All Students	(1)	(2)	(3)
Bilingual teacher	-0.170	-0.158	0.032
	(0.104)	(0.137)	(0.152)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.469	0.469	0.737
Observations	8,937	8,937	8,937
Panel B: Native Students			
Bilingual teacher	-0.301	-0.269	0.106
-	(0.217)	(0.252)	(0.247)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.438	0.438	0.722
Observations	5,952	5,952	5,952
Panel C: Foreign Origin Students			
Bilingual teacher	-0.205^{**}	-0.207	0.065
	(0.096)	(0.128)	(0.224)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.502	0.501	0.746
Observations	2,985	2,985	2,985

Panel A: All Students	(1)	(2)	(3)
Language family (Ref.: German)			
Slavic	0.222^{**}	0.471^{**}	* 0.426**
	(0.090)	(0.143)	(0.205)
Romanic	0.426^{***}	0.282**	0.329
	(0.100)	(0.140)	(0.343)
Others	0.248	0.151	0.587^{**}
	(0.190)	(0.221)	(0.293)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.377	0.378	0.661
Observations	$7,\!351$	7,351	7,351
Panel B: Native Students			
Language family (Ref.: German)			
Slavic	0.273^{***}	0.529***	* 0.462*
	(0.096)	(0.169)	(0.254)
Romanic	0.410^{**}	0.330	0.306
	(0.207)	(0.225)	(0.570)
Others	0.393^{*}	0.503^{*}	0.917^{**}
	(0.219)	(0.278)	(0.340)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.362	0.363	0.649
Observations	4,812	4,812	4,812
Panel C: Foreign Origin Students			
Language family (Ref.: German)			
Slavic	0.183	0.440^{**}	0.687^{**}
	(0.149)	(0.187)	(0.318)
Romanic	0.420^{**}	0.086	0.336
	(0.178)	(0.214)	(0.395)
Others	-0.005	-0.087	0.476
	(0.269)	(0.328)	(0.353)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R ²	0.394	0.393	0.655
Observations	2,539	2,539	2,539

Table A17: Effect of Teacher's Native Languageon Reading Comprehension

Table A18: Effect of Foreign Origin Teacher on
Reading Comprehension Controlling For
Aspects of Teacher's Career Choice

Panel A: All Students	(1)	(2)	(3)
Foreign origin teacher	0.175^{**}	0.229***	0.273^{*}
	(0.069)	(0.070)	(0.140)
Aspects of teacher's career choice		0.059	0.050
Enjoy teaching		-0.053	-0.056
Prestige of teachers		(0.030)	(0.007)
		(0.027)	(0.038)
Enjoy subject		-0.050	0.008
		(0.041)	(0.064)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.377	0.378	0.660
Observations	7,351	7,351	7,351
Panel B: Native Students			
Dension esisin tereben	0.050**	0.000***	0.001
Foreign origin teacher	(0.259)	(0.290)	(0.291)
Aspects of teacher's career choice	(0.105)	(0.103)	(0.211)
Enjoy teaching	-0.151^{**}	-0.112^{*}	-0.101
Enjoy touoning	(0.063)	(0.062)	(0.077)
Prestige of teachers	0.037	0.031	0.028
0	(0.027)	(0.032)	(0.048)
Enjoy subject	-0.053	-0.082^{*}	-0.036
	(0.045)	(0.049)	(0.081)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.363	0.363	0.648
Observations	4,812	4,812	4,812
Panel C: Foreign Origin Students			
Foreign origin too shore	0.120	0.111	0.200*
roreign origin teacher	(0.129)	(0.098)	(0.388)
Aspects of teacher's career choice	(0.031)	(0.038)	(0.201)
Enjoy teaching	0.049	0.137^{*}	0.043
Enjoy touoning	(0.068)	(0.077)	(0.141)
Prestige of teachers	0.027	-0.003	0.020
0	(0.031)	(0.043)	(0.061)
Enjoy subject	-0.025	$-0.060^{-0.060}$	0.036
	(0.056)	(0.064)	(0.110)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.394	0.393	0.654
Observations	2,539	2,539	2,539

Foreign origin teacher 0.175** 0.185*** 0.227* Teacher's cultural orientation 0.0371 0.0371 Relevance of multicultural ideals -0.023 0.011 Consideration of differences in class 0.012 -0.039 Consideration of differences in class 0.122*** 0.0361 Resolving disputes through similarities 0.0461 0.0757 Problem if students do not adapt to German culture -0.041 -0.023 Objective to promote things in common 0.066 0.0101 Cultural diversity in teacher training -0.043 -0.0201 Student CPE Yes Yes Yes Student FE No No Yes Student FE No No Yes Foreign origin teacher 0.236** 0.255** 0.255 Consideration of differences in class 0.001 -0.043 0.0660 Obeer origin teacher 0.236** 0.255** 0.255* 0.255* Foreign origin teacher 0.0661 -0.043 -0.0647 Problem if studen	Panel A: All Students	(1)	(2)	(3)
Teacher's cultural orientation (0.019) (0.011) (0.131) Relevance of multicultural ideals -0.023 0.111 Relevance of multicultural ideals -0.023 0.111 Resolving disputes through similarities 0.012 -0.039 Problem if students do not adapt to German culture -0.023 0.012 Objective to promote things in common 0.006 0.0023 (0.036) Objective to promote things in common 0.006 0.0006 0.0010 Class FE Yes Yes Yes Yes Student Controls No Yes Yes Yes Adjusted R ² 0.377 0.378 0.660 Observations 7.351 7.351 7.351 Panel B: Native Students 0.024 0.662 0.229^{**} Feacher's cultural orientation (0.047) (0.047) (0.047) Relevance of multicultural ideals 0.024 0.662 0.219^{**} Consideration of differences in class 0.0061 -0.064^{**} $0.$	Foreign origin teacher	0.175^{**}	0.185^{***}	0.227^{*}
Relevance of multicultural ideals -0.023 0.111 Consideration of differences in class 0.021 -0.030 Resolving disputes through similarities 0.122^{+++} 0.036 Problem if students do not adapt to German culture 0.022^{+++} 0.036 Objective to promote things in common 0.006 0.000 Cultural diversity in teacher training -0.043 -0.020 Class FE Yes Yes Yes Student controls Yes Yes Yes Teacher controls Yes Yes Yes Student FE No No No Yes Foreign origin teacher 0.236^{++} 0.237^{++} 0.230^{++} Relevance of multicultural ideals 0.066^{+} 0.041^{++} 0.020^{++} Consideration of differences in class 0.001^{+} -0.104^{+} 0.023^{++} Resolving disputes through similarities 0.156^{++} 0.131^{++} 0.023^{++} Consideration of differences in class 0.006^{++} -0.008^{++} -0.008^{++}	Teacher's cultural orientation	(0.009)	(0.071)	(0.137)
Consideration of differences in class (0.053) (0.039) Resolving disputes through similarities (0.122) (0.036) (0.067) Problem if students do not adapt to German culture (0.046) (0.075) Objective to promote things in common (0.043) (0.060) Cultural diversity in teacher training (0.040) (0.043) (0.060) Cultural diversity in teacher training (0.040) (0.040) (0.062) Class FE Yes Yes Yes Yes Student controls No No Yes Yes Adjusted R ² 0.377 0.378 0.660 Observations (0.116) (0.113) (0.220) Teacher is cultural orientation (0.047) (0.048) (0.220) Resolving disputes through similarities 0.156** 0.255* 0.257** 0.257 Consideration of differences in class 0.001 -0.047* -0.104 (0.125) Consideration of differences in class 0.060 0.057* (0.043) (0.029) Objective to promote t	Relevance of multicultural ideals		-0.023	0.111
Construction of uniferences in class (0.036) (0.067) Resolving disputes through similarities (0.046) (0.075) Problem if students do not adapt to German culture -0.041^+ -0.021 Objective to promote things in common 0.006 0.012^- Objective to promote things in common 0.006 0.0100^- Class FE Yes Yes Yes Student controls No Yes Yes Student controls No Yes Yes Student FE No No Yes Adjusted R ² 0.377 0.378 0.660 Observations 7.351 7.351 7.351 Precisin origin teacher $(0.160)^+$ $(0.113)^ (0.220)^+$ Resolving disputes through similarities 0.060^+ $(0.125)^ 0.236^+$ Consideration of differences in class $(0.007)^ 0.003^ 0.001^-$ Consideration of differences in class $0.000^ 0.053^ 0.023^+$ 0.023^+ 0.023^+ Prob	Consideration of differences in class		(0.054) 0.012	(0.099) -0.039
Resolving disputes through similarities 0.122*** 0.051 Problem if students do not adapt to German culture 0.0461 0.0461 0.035) Objective to promote things in common 0.066 0.010 0.006 0.010 Cultural diversity in teacher training -0.043 -0.020 (0.040) (0.062) Class FE Yes Yes Yes Yes Yes Yes Student controls Yes Yes Yes Yes Yes Yes Adjusted R ² 0.377 0.378 0.660 0.0220 Deservations 7.351 7.351 7.351 Panel B: Native Students 0.024 0.062 0.219* Relevance of multicultural ideals 0.024 0.062 0.219* Resolving disputes through similarities 0.156** 0.155** 0.055 Consideration of differences in class 0.001 -0.064** 0.023 Objective to promote things in common 0.053 0.0481 0.0471 0.0431 Objective to promote things in common <td< td=""><td>Consideration of universities in class</td><td></td><td>(0.012)</td><td>(0.067)</td></td<>	Consideration of universities in class		(0.012)	(0.067)
Problem if students do not adapt to German culture (0.046) (0.072) Objective to promote things in common 0.006 0.010 Cultural diversity in teacher training -0.043 -0.020 Class FE Yes Yes Yes Student controls No Yes Yes Student controls No Yes Yes Adjusted R ² 0.377 0.378 0.660 Observations 7.351 7.351 7.351 Foreign origin teacher (0.116) (0.113) (0.225) Consideration of differences in class 0.004 (0.066) (0.125) Consideration of differences in class 0.0047 (0.048) (0.083) Resolving disputes through similarities 0.156^{++	Resolving disputes through similarities		0.122***	0.051
Toolean it students to tot a dapt to German Cutute -0.011 -0.021 Objective to promote things in common 0.006 0.0101 Clutural diversity in teacher training -0.043 -0.0201 Class FE Yes Yes Yes Student controls Yes Yes Yes Student controls No Yes Yes Adjusted R ² 0.377 0.378 0.6600 Observations $7,351$ $7,351$ $7,351$ Panel B: Native Students (0.116) (0.113) (0.225) Consideration of differences in class 0.001 -0.047 -0.104 Resolving disputes through similarities 0.156^{+*} 0.154^{+**} 0.052 Objective to promote things in common 0.033 0.048 0.0051 Objective to promote things in common 0.053 0.048 0.0051 Objective to promote things in common 0.0541 0.0071 -0.037 Objective to promote things in common 0.0541 0.0071 0.0171	Problem if students do not adapt to Corman culture		(0.046)	(0.075)
Objective to promote things in common 0.006 0.010 Cultural diversity in teacher training -0.043 -0.020 Class FE Yes Yes Yes Student controls Yes Yes Yes Student FE No No Yes Adjusted R2 0.377 0.378 0.6660 Observations 7,351 7,351 7,351 Panel B: Native Students - 0.116 (0.113) (0.220) Teacher controls 0.024 0.066 0.219* Observations 0.044 0.062 0.219* Consideration of differences in class 0.001 -0.047 -0.104 (0.047) (0.048) (0.068) (0.069) Problem if students do not adapt to German culture -0.066** -0.004** -0.003* Objective to promote things in common 0.053 0.044 0.005 Class FE Yes Yes Yes Student controls Yes Yes Yes Colstil erity in teac	r toblem il students do not adapt to German culture		(0.023)	(0.036)
Cultural diversity in teacher training $(0.043) = (0.043) = (0.066) \\ (0.040) = (0.062) \\ (0.040) = (0.062) \\ (0.040) = (0.062) \\ (0.062) = (0.062) \\ (0.062) = (0.062) \\ (0.062) = (0.062) \\ (0.062) = (0.062) \\ (0.062) = (0.062) \\ (0.062) = (0.062) \\ (0.062) = (0.062) \\ (0.062) = (0.062) \\ (0.062) = (0.062) \\ (0.063) = (0.062) \\ (0.066) = (0.125) \\ (0.066) = (0.025) \\ (0.068) = (0.064) \\ (0.063) = (0.064) \\ (0.061) = (0.051) \\ (0.051) = (0.051) \\ (0.051) = (0.051) \\ (0.074) \\ (0.074) \\ (0.051) = (0.051) \\ (0.074) \\ (0.074) \\ (0.074) \\ (0.074) \\ (0.074) \\ (0.074) \\ (0.074) \\ (0.074) \\ (0.074) \\ (0.051) = (0.051) \\ (0.074) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.075) \\ (0.07$	Objective to promote things in common		0.006	0.010
Class FE Yes Yes Yes Student controls Yes Yes Yes Student controls Yes Yes Yes Student Controls Yes Yes Yes Student FE No No Yes Adjusted R ² 0.377 0.378 0.660 Observations 7,351 7,351 7,351 Panel B: Native Students (0.116) (0.113) (0.220) Teacher's cultural orientation (0.069) (0.066) (0.125) Consideration of differences in class (0.061) (0.077) (0.043) Objective to promote things in common 0.063 0.0448 0.0052 Objective to promote things in common (0.054) (0.043) (0.043) Objective to promote things in common (0.054) (0.047) (0.0448 Objective to promote things in common (0.054) (0.047) (0.043) Objective to promote things in common (0.054) (0.047) (0.0448) Outard diversity in teacher training <t< td=""><td>Cultural discovity in teacher training</td><td></td><td>(0.043)</td><td>(0.066)</td></t<>	Cultural discovity in teacher training		(0.043)	(0.066)
Class FE Yes Yes Yes Yes Student controls No Yes Yes Yes Student FE No No Yes Yes Adjusted R ² 0.377 0.378 0.6660 Observations 7,351 7,351 7,351 Panel B: Native Students 0.236** 0.255** 0.225 Foreign origin teacher 0.236** 0.062 0.219* Consideration of differences in class 0.001 -0.017 -0.104 Resolving disputes through similarities 0.666* 0.154*** 0.052 Problem if students do not adapt to German culture -0.0667* -0.064** -0.008 Objective to promote things in common 0.053 0.048 0.005 (0.047) (0.047) (0.076) Cultural diversity in teacher training -0.073 -0.100** -0.037 Fos Yes Yes </td <td>Cultural diversity in teacher training</td> <td></td> <td>(0.043)</td> <td>(0.062)</td>	Cultural diversity in teacher training		(0.043)	(0.062)
Student controls Yes Yes Yes Yes Iteacher controls No No No Yes Student FE No No Yes Adjusted R ² 0.377 0.378 0.660 Observations 7,351 7,351 7,351 Panel B: Native Students 0.236** 0.255** 0.255 Teacher's cultural orientation (0.116) (0.113) (0.220) Teacher's cultural orientation (0.069) (0.066) (0.125) Consideration of differences in class 0.001 -0.047 -0.104 Consideration of differences in class 0.016** 0.154**** 0.052 Consideration of differences in class 0.0600 (0.057) (0.093) Problem if students do not adapt to German culture -0.066** -0.064** -0.008 Objective to promote things in common 0.053 0.048 0.0051 (0.076) Cultural diversity in teacher training -0.073 -0.100** -0.037 Cultural diversity in teacher training -0.051 <td>Class FE</td> <td>Yes</td> <td>Yes</td> <td>Yes</td>	Class FE	Yes	Yes	Yes
Teacher controls No Yes Yes Student FE No No Yes Adjusted R ² 0.377 0.378 0.660 Observations 7,351 7,351 7,351 Panel B: Native Students (0.116) (0.113) (0.220) Teacher's cultural orientation (0.069) (0.066) (0.125) Consideration of differences in class (0.001) -0.047 -0.104 (0.047) (0.0448) (0.083) Resolving disputes through similarities (0.060) (0.070) (0.0448) (0.033) Objective to promote things in common 0.063 0.044 0.0071 (0.043) Objective to promote things in common (0.051) (0.076) (0.076) Cultural diversity in teacher training -0.073 -0.104* -0.008 Cultural diversity in teacher training -0.073 -0.104** -0.008 (0.074) (0.041/) (0.074) Class FE Yes Yes Yes Yes Yes Yes Yes Yes Yes <t< td=""><td>Student controls</td><td>Yes</td><td>Yes</td><td>Yes</td></t<>	Student controls	Yes	Yes	Yes
Student FL 10 10 105 105 Adjusted R ² Observations 0.377 0.378 0.660 Observations 7,351 7,351 7,351 Panel B: Native Students (0.116) (0.113) (0.220) Foreign origin teacher 0.236** 0.255** 0.255 Consideration of differences in class 0.001 -0.047 -0.104 (0.047) (0.048) (0.059) (0.069) (0.051) Consideration of differences in class 0.011 -0.047 -0.104 (0.047) (0.048) (0.083) (0.051) (0.077) Problem if students do not adapt to German culture -0.064** -0.064* -0.064* Objective to promote things in common 0.053 0.048 0.005 Cultural diversity in teacher training -0.073 -0.100** -0.037 Class FE Yes Yes Yes Yes Student FE No No Yes Yes Adjusted R ² 0.363 0.364 0.648 </td <td>Teacher controls Student FF</td> <td>No No</td> <td>Yes</td> <td>Yes</td>	Teacher controls Student FF	No No	Yes	Yes
Adjusted R ² 0.377 0.378 0.630 Observations 7,351 7,351 7,351 Panel B: Native Students (0.116) (0.113) (0.220) Teacher's cultural orientation (0.069) (0.062) 0.219* Relevance of multicultural ideals 0.024 0.062 0.219* Consideration of differences in class 0.001 -0.047 -0.104 (0.060) (0.060) (0.067) (0.093) Problem if students do not adapt to German culture -0.064* -0.064* -0.063 Objective to promote things in common 0.053 0.048 0.005 Cultural diversity in teacher training -0.073 -0.100** -0.063 Class FE Yes Yes Yes Yes Student controls Yes Yes Yes Yes Student FE No No Yes Yes Foreign origin teacher 0.150* 0.159 0.331 Objective to promote things in common 0.603 0.363 0.364 0.648 Objective to promote things No Yes Ye	Student FE	NO	NO	Tes
Observations 7,351 7,351 7,351 Panel B: Native Students	Adjusted R ²	0.377	0.378	0.660
Panel B: Native Students Foreign origin teacher 0.236** 0.255** 0.255 (0.116) (0.113) (0.220) Teacher's cultural orientation (0.069) (0.066) (0.125) Consideration of differences in class 0.001 -0.047 -0.104 (0.047) (0.060) (0.056) (0.125) Consideration of differences in class 0.001 -0.047 -0.104 (0.047) (0.048) (0.083) 1.54*** 0.052 Problem if students do not adapt to German culture -0.060** -0.064** -0.069 Objective to promote things in common 0.053 0.048 0.0051 Cultural diversity in teacher training -0.073 -0.100** -0.037 Class FE Yes Yes Yes Yes Student controls Yes Yes Yes Yes Student FE No No Yes Yes Student controls Yes Yes Yes Yes Adjusted R ² 0.363 0	Observations	7,351	7,301	7,351
Foreign origin teacher 0.236^{**} 0.255^{**} 0.255 reacher's cultural orientation (0.11) (0.220) Relevance of multicultural ideals 0.024 0.062 0.219^* Consideration of differences in class 0.001 -0.047 -0.104 (0.047) (0.048) (0.083) Resolving disputes through similarities 0.156^{**} 0.154^{***} 0.052 Objective to promote things in common 0.060^* -0.068^* -0.008 Objective to promote things in common 0.053 0.048 0.005 Cultural diversity in teacher training -0.073 -0.007^* -0.037 Cultural diversity in teacher training 0.053 0.48 0.005 Student controls Yes Yes Yes Yes Student controls Yes Yes Yes Yes Adjusted R ² 0.363 0.364 0.643 0.629 0.331 Observations 0.041 -0.050 0.215^* 0.159^* 0.159^*	Panel B: Native Students			
(0.116) (0.113) (0.220) Relevance of multicultural ideals 0.024 0.062 0.219* Consideration of differences in class 0.001 -0.047 -0.104 (0.069) (0.068) (0.033) Resolving disputes through similarities 0.156** 0.154*** 0.052 Problem if students do not adapt to German culture -0.060** -0.064** -0.008 Objective to promote things in common 0.053 0.048 0.0051 (0.074) Cultural diversity in teacher training -0.073 -0.104** -0.037 (0.054) (0.047) (0.074) Class FE Yes	Foreign origin teacher	0.236**	0.255**	0.255
Relevance of multicultural ideals 0.024 0.062 0.219* Relevance of multicultural ideals (0.069) (0.066) (0.125) Consideration of differences in class 0.001 -0.047 -0.104 Resolving disputes through similarities 0.156** 0.154*** 0.052 Problem if students do not adapt to German culture -0.060** -0.064** -0.008 Objective to promote things in common 0.053 0.048 0.005 Cultural diversity in teacher training -0.073 -0.100** -0.037 (0.054) (0.047) (0.074) (0.047) (0.074) Class FE Yes Yes Yes Yes Yes Student controls Yes Yes Yes Yes Yes Adjusted R ² 0.363 0.364 0.648 0.515) (0.215) Teacher controls 4,812 4,812 4,812 4,812 Panel C: Foreign Origin Students -0.036 (0.047) (0.215) Teacher's cultural orientation -0.041 0.041 0.051 Relevance of multicultural ideals 0.041	The show's conference of second strength	(0.116)	(0.113)	(0.220)
Intervation of differences in class (0.069) (0.067) (0.125) Consideration of differences in class 0.001 -0.047 -0.104 Resolving disputes through similarities 0.156^{**} 0.154^{***} 0.052 Problem if students do not adapt to German culture -0.066^{**} -0.008^{**} -0.008 Objective to promote things in common 0.053 0.048 0.0051 Cultural diversity in teacher training -0.073 -0.100^{**} -0.037 Cultural diversity in teacher training (0.042) (0.047) (0.074) Class FE Yes Yes Yes Yes Student controls Yes Yes Yes Yes Student FE No No Yes Yes Adjusted R ² 0.363 0.364 0.648 Observations 4.812 4.812 4.812 Panel C: Foreign Origin Students (0.064) (0.048) (0.0215) Consideration of differences in class 0.041 -0.050 (0.058)	Relevance of multicultural ideals	0.024	0.062	0.219^{*}
Consideration of differences in class 0.001 -0.047 -0.104 Resolving disputes through similarities 0.156^{++} 0.154^{+++} 0.083 Problem if students do not adapt to German culture -0.060^{++} -0.066^{++} -0.008 Objective to promote things in common 0.053 0.048 0.0051 Cultural diversity in teacher training -0.073 -0.100^{++} -0.037 Cultural diversity in teacher training (0.042) (0.047) (0.074) Class FE Yes Yes Yes Yes Student controls No Yes Yes Yes Student controls No No Yes Yes Observations 4,812 4,812 4,812 Panel C: Foreign Origin Students 0.041 -0.050 -0.050 Teacher's cultural orientation 0.041 -0.050 -0.050 Relevance of multicultural ideals 0.041 -0.028 -0.052 Consideration of differences in class 0.041 0.041 0.051		(0.069)	(0.066)	(0.125)
(0.047) (0.048) (0.083) Resolving disputes through similarities 0.156^{+++} 0.154^{+++} 0.052 (0.060) (0.057) (0.093) Problem if students do not adapt to German culture -0.060^{++} -0.064^{++} -0.008 (0.029) (0.028) (0.043) (0.043) (0.029) (0.028) (0.043) $Objective to promote things in common0.0530.0440.005(0.042)(0.076)Cultural diversity in teacher training-0.073-0.100^{++}-0.037(0.054)(0.047)(0.074)(0.074)Class FEYesYesYesStudent controlsYesYesYesStudent FENoNoYesAdjusted R20.3630.3640.648Observations4,8124,8124,812Panel C: Foreign Origin Students(0.090)(0.106)(0.215)Teacher control se(0.066)(0.092)(0.158)Consideration of differences in class0.041-0.050-0.050(0.066)(0.072)(0.074)(0.074)Problem if students do not adapt to German culture-0.017-0.028-0.052(0.043)(0.045)(0.014)0.0410.0210.010(0.051)(0.060)(0.071)(0.074)(0.074)Problem if students do not adapt to German culture-0.017-0.028-0.052(0.043)$	Consideration of differences in class	0.001	-0.047	-0.104
Resolving disputes through similarities 0.156^{++} 0.154^{++} 0.052 Problem if students do not adapt to German culture (0.060) (0.077) (0.093) Objective to promote things in common 0.053 0.048 0.005 Cultural diversity in teacher training -0.073 -0.100^{+*} -0.037 Class FE Yes Yes Yes Student controls Yes Yes Yes Student controls Yes Yes Yes Student FE No No Yes Adjusted R ² 0.363 0.364 0.648 Observations $4,812$ $4,812$ $4,812$ $4,812$ Panel C: Foreign Origin Students (0.090) (0.160) (0.215) Teacher's cultural orientation (0.066) (0.072) (0.158) Consideration of differences in class 0.014 0.051 (0.064) (0.068) (0.114) Resolving disputes through similarities 0.014 0.021 0.012 (0.074) (0.074) (0.074) Objective to promote things in common		(0.047)	(0.048)	(0.083)
Problem if students do not adapt to German culture $(0.000)^*$ -0.064^{**} -0.008 Objective to promote things in common 0.053 0.048 0.005 Cultural diversity in teacher training -0.073 -0.100^{**} -0.037 Cultural diversity in teacher training -0.073 -0.100^{**} -0.037 Class FE Yes Yes Yes Yes Yes Student controls Yes Yes Yes Yes Yes Adjusted R ² 0.363 0.364 0.648 0.648 Observations 4,812 4,812 4,812 4,812 Panel C: Foreign Origin Students (0.090) (0.106) (0.215) Teacher's cultural orientation (0.086) (0.092) (0.158) Relevance of multicultural ideals 0.041 -0.051 (0.051) Consideration of differences in class 0.041 0.041 0.051 (0.066) (0.072) (0.114) 0.051 Problem if students do not adapt to German culture -0.017 -0.052 <td>Resolving disputes through similarities</td> <td>(0.156^{**})</td> <td>(0.154^{***})</td> <td>(0.052)</td>	Resolving disputes through similarities	(0.156^{**})	(0.154^{***})	(0.052)
Objective to promote things in common (0.029) (0.028) (0.043) Objective to promote things in common 0.053 0.048 0.005 Cultural diversity in teacher training -0.073 -0.100^{**} -0.037 Class FE Yes Yes Yes Yes Student controls Yes Yes Yes Student FE No No Yes Adjusted R ² 0.363 0.364 0.648 Observations $4,812$ $4,812$ $4,812$ Panel C: Foreign Origin Students -0.050 -0.050 -0.050 Consideration of differences in class 0.041 -0.050 -0.050 Consideration of differences in class 0.041 0.021 0.010 Problem if students do not adapt to German culture -0.017 -0.028 -0.052 (0.072) (0.073) (0.066) (0.074) (0.074) Panel C: Foreign Origin Students -0.017 -0.028 -0.050 Consideration of differences in class	Problem if students do not adapt to German culture	-0.060^{**}	-0.064^{**}	-0.008
Objective to promote things in common 0.053 0.048 0.005 Cultural diversity in teacher training (0.042) (0.076) (0.076) Cultural diversity in teacher training (0.054) (0.074) (0.074) Class FE Yes Yes Yes Yes Student controls No Yes Yes Student FE No No Yes Adjusted R ² 0.363 0.364 0.648 Observations $4,812$ $4,812$ $4,812$ Panel C: Foreign Origin Students (0.090) (0.106) (0.215) Teacher's cultural orientation (0.086) (0.092) (0.158) Consideration of differences in class 0.041 0.061 0.051 Consideration of differences in class 0.041 0.021 0.010 Problem if students do not adapt to German culture -0.017 -0.028 -0.052 (0.043) (0.043) (0.045) (0.074) Objective to promote things in common -0.017		(0.029)	(0.028)	(0.043)
	Objective to promote things in common	0.053	0.048	0.005
Control diversity in teacher training -0.054 -0.047 -0.037 Class FE Yes Yes Yes Yes Student controls No Yes Yes Teacher controls No No Yes Student FE No No Yes Adjusted R ² 0.363 0.364 0.648 Observations 4,812 4,812 4,812 Panel C: Foreign Origin Students 0.000 (0.160) (0.215) Teacher's cultural orientation 0.041 -0.050 -0.050 Relevance of multicultural ideals 0.041 -0.051 0.016 Consideration of differences in class 0.041 0.021 0.010 Consideration of differences in class 0.014 0.021 0.010 Problem if students do not adapt to German culture -0.017 -0.028 -0.052 (0.072) (0.073) (0.045) (0.074) 0.0114 Objective to promote things in common -0.107 -0.028 -0.052 (0.072)	Cultural diversity in teacher training	(0.042) 0.073	(0.051) 0.100**	(0.076) 0.037
Class FE Yes Yes Yes Yes Student controls No Yes Yes Yes Teacher controls No No Yes Yes Student FE No No Yes Yes Adjusted R ² 0.363 0.364 0.648 Observations 4,812 4,812 4,812 Panel C: Foreign Origin Students 0.059 0.331 (0.090) (0.106) (0.215) Teacher's cultural orientation (0.086) (0.092) (0.158) 0.051 Consideration of differences in class 0.041 0.041 0.051 (0.064) (0.068) (0.114) 0.051 Resolving disputes through similarities 0.017 -0.028 -0.052 (0.043) (0.045) (0.074) 0.010 Objective to promote things in common -0.107 -0.012 -0.006 (0.051) (0.060) (0.100) (0.114) Objective to promote things in common -0.017 -0.012 -0.006<	Cultural diversity in teacher training	(0.054)	(0.047)	(0.074)
Student controls Yes Yes Yes Yes Teacher controls No No Yes Yes Student FE No No Yes Adjusted R ² 0.363 0.364 0.648 Observations 4,812 4,812 4,812 Panel C: Foreign Origin Students (0.090) (0.166) (0.215) Teacher's cultural orientation (0.086) (0.092) (0.158) Consideration of differences in class 0.041 -0.050 -0.050 Consideration of differences in class 0.041 0.021 0.010 Resolving disputes through similarities 0.014 0.021 0.010 Problem if students do not adapt to German culture -0.017 -0.028 -0.052 (0.043) (0.045) (0.074) (0.0114) Objective to promote things in common -0.107 -0.112 -0.006 (0.051) (0.060) (0.100) (0.151) Cultural diversity in teacher training 0.094^*	Class FE	Yes	Yes	Yes
Teacher controls No Yes Yes Student FE No No Ves Adjusted R ² 0.363 0.364 0.648 Observations 4,812 4,812 4,812 Panel C: Foreign Origin Students (0.990) (0.159) 0.331 Foreign origin teacher 0.150^* 0.159 0.331 Relevance of multicultural ideals 0.041 -0.050 -0.050 Consideration of differences in class 0.041 -0.050 -0.051 Resolving disputes through similarities 0.041 0.041 0.051 Problem if students do not adapt to German culture -0.017 -0.028 -0.052 (0.043) (0.043) (0.045) (0.074) Objective to promote things in common -0.107 -0.112 -0.006 (0.051) (0.060) (0.114) (0.051) (0.060) (0.115) Cultural diversity in teacher training 0.094^* 0.094 0.031 (0.051) (0.060) (0.100) (0.151) (0.060) (0.100)	Student controls	Yes	Yes	Yes
Durdent FE 1000 1000	Teacher controls Student FE	No No	Yes	Yes
Adjusted \mathbb{R}^2 0.363 0.364 0.648 Observations 4,812 4,812 4,812 Panel C: Foreign Origin Students (0.090) (0.159 0.331 Foreign origin teacher 0.150* 0.159 0.331 (0.090) (0.106) (0.215) Teacher's cultural orientation (0.090) (0.106) (0.215) Relevance of multicultural ideals 0.041 -0.050 -0.050 Consideration of differences in class 0.041 0.041 0.051 Consideration of differences in class 0.041 0.021 0.010 Resolving disputes through similarities 0.014 0.021 0.010 Problem if students do not adapt to German culture -0.017 -0.028 -0.052 (0.043) (0.043) (0.045) (0.074) Objective to promote things in common -0.107 -0.112 -0.006 (0.051) (0.060) (0.100) (0.100) (0.151) Cultural diversity in teacher training 0.094* 0.094 0.031 (0.051) (0.060) (0.100) (0.100) <t< td=""><td>Student FE</td><td>NO</td><td>110</td><td>165</td></t<>	Student FE	NO	110	165
Observations $4,812$ $4,612$ $4,612$ $4,612$ $4,612$ $4,612$ Panel C: Foreign Origin Students End of the second state of the se	Adjusted R ²	0.363	0.364	0.648
Panel C: Foreign Origin Students Foreign origin teacher 0.150^* 0.159 0.331 Foreign origin teacher (0.090) (0.106) (0.215) Teacher's cultural orientation (0.090) (0.106) (0.215) Teacher's cultural orientation (0.086) (0.092) (0.158) Consideration of differences in class 0.041 0.041 0.051 Consideration of differences in class 0.014 0.021 0.010 Resolving disputes through similarities 0.014 0.021 0.010 Problem if students do not adapt to German culture -0.017 -0.028 -0.052 (0.043) (0.045) (0.074) 0.094^* 0.094 Objective to promote things in common -0.107 -0.112 -0.006 (0.072) (0.075) (0.115) Cultural diversity in teacher training 0.094^* 0.094 0.031 (0.51) (0.060) (0.100) Class FE Yes Yes Student controls <t< td=""><td>Observations</td><td>4,012</td><td>4,012</td><td>4,012</td></t<>	Observations	4,012	4,012	4,012
Foreign origin teacher 0.150^* 0.159 0.331 Teacher's cultural orientation (0.090) (0.106) (0.215) Teacher's cultural orientation 0.041 -0.050 -0.050 Relevance of multicultural ideals 0.041 -0.041 0.051 Consideration of differences in class 0.041 0.041 0.051 Consideration of differences in class 0.041 0.041 0.051 Resolving disputes through similarities 0.014 0.021 0.010 Problem if students do not adapt to German culture -0.017 -0.028 -0.052 Objective to promote things in common -0.107 -0.112 -0.006 Cultural diversity in teacher training 0.094^* 0.094 0.031 Class FE Yes Yes Yes Student controls Yes Yes Yes Teacher controls No Yes Yes Adjusted R ² 0.394 0.393 0.653 Observations 2.539 2.539 2.539 <td>Panel C: Foreign Origin Students</td> <td></td> <td></td> <td></td>	Panel C: Foreign Origin Students			
	Foreign origin teacher	0.150*	0.159	0.331
Relevance of multicultural ideals 0.041 -0.050 -0.050 Relevance of multicultural ideals (0.086) (0.092) (0.158) Consideration of differences in class 0.041 0.041 0.051 Resolving disputes through similarities 0.014 0.021 0.010 Problem if students do not adapt to German culture -0.077 -0.028 -0.052 (0.043) (0.045) (0.074) $0.074)$ Objective to promote things in common -0.107 -0.112 -0.006 (0.051) (0.060) (0.114) Class FE Yes Yes Yes Student controls Yes Yes Yes Teacher controls Yes Yes Yes Student FE No No Yes Adjusted R ² 0.394 0.393 0.653 Observations 2.539 2.539 2.539	Teacher's cultural orientation	(0.090)	(0.106)	(0.215)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Relevance of multicultural ideals	0.041	-0.050	-0.050
$\begin{array}{ccccc} {\rm Consideration of differences in class} & 0.041 & 0.041 & 0.051 \\ & & (0.064) & (0.068) & (0.114) \\ {\rm Resolving disputes through similarities} & 0.014 & 0.021 & 0.010 \\ & & (0.066) & (0.072) & (0.114) \\ {\rm Problem if students do not adapt to German culture} & -0.017 & -0.028 & -0.052 \\ & & (0.043) & (0.045) & (0.074) \\ {\rm Objective to promote things in common} & -0.107 & -0.112 & -0.006 \\ & & (0.072) & (0.075) & (0.115) \\ {\rm Cultural diversity in teacher training} & 0.094^{*} & 0.094 & 0.031 \\ & & (0.051) & (0.060) & (0.100) \\ {\rm Class FE} & {\rm Yes} & {\rm Yes} & {\rm Yes} \\ {\rm Student controls} & {\rm Yes} & {\rm Yes} & {\rm Yes} \\ {\rm Student FE} & {\rm No} & {\rm No} & {\rm Yes} \\ {\rm Adjusted R}^2 & 0.394 & 0.393 & 0.653 \\ {\rm Observations} & 2.539 & 2.539 & 2.539 \end{array}$		(0.086)	(0.092)	(0.158)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Consideration of differences in class	0.041	0.041	0.051
Resolving disputes through similarities 0.014 0.021 0.010 Problem if students do not adapt to German culture -0.017 -0.028 -0.052 (0.043) (0.045) (0.074) 0.014 Objective to promote things in common -0.017 -0.028 -0.052 (0.072) (0.074) (0.074) (0.074) (0.074) Objective to promote things in common -0.107 -0.112 -0.006 (0.072) (0.075) (0.115) Cultural diversity in teacher training 0.094^* 0.094 0.031 (0.051) (0.060) (0.100) Class FE Yes Yes Yes Student controls Yes Yes Yes Student FE No No Yes Adjusted R ² 0.394 0.393 0.653 Observations 2.539 2.539 2.539	Deceluing disputes through similarities	(0.064)	(0.068)	(0.114)
Problem if students do not adapt to German culture -0.017 -0.052 Objective to promote things in common -0.107 -0.112 -0.006 (0.072) (0.075) (0.115) Cultural diversity in teacher training 0.094^* 0.094 0.031 Class FE Yes Yes Yes Student controls Yes Yes Yes Teacher controls No Yes Yes Student FE No No Yes Adjusted R ² 0.394 0.393 0.653 Observations 2.539 2.539 2.539	Resolving disputes through similarities	(0.014)	(0.021)	(0.114)
$\begin{array}{cccc} & (0.043) & (0.045) & (0.074) \\ & \text{Objective to promote things in common} & & -0.107 & -0.112 & -0.006 \\ & (0.072) & (0.075) & (0.115) \\ & \text{Cultural diversity in teacher training} & & 0.094^* & 0.094 & & 0.031 \\ & (0.051) & (0.060) & (0.100) \\ & \text{Class FE} & & Yes & Yes \\ & Yes & Yes & Yes & Yes \\ & Student controls & & Yes & Yes \\ & Teacher controls & & Yes & Yes \\ & Student FE & & No & Yes & Yes \\ & Student FE & & No & No & Yes \\ & \end{tabular}$	Problem if students do not adapt to German culture	-0.017	-0.028	-0.052
Objective to promote things in common -0.107 -0.112 -0.006 (0.072) (0.075) (0.115) Cultural diversity in teacher training 0.094^* 0.094 0.031 (0.051) (0.060) (0.100) Class FE Yes Yes Yes Student controls Yes Yes Yes Teacher controls No Yes Yes Student FE No No Yes Adjusted R ² 0.394 0.393 0.653 Observations 2.539 2.539 2.539		(0.043)	(0.045)	(0.074)
$ \begin{array}{c} (0.072) & (0.073) & (0.115) \\ (0.072) & 0.094 & 0.094 & 0.031 \\ (0.051) & (0.060) & (0.100) \\ (0.051) & (0.060) & (0.100) \\ (0.051) & (0.060) & (0.100) \\ (0.051) & (0.060) & (0.100) \\ (0.060) & (0.100) \\ (0.051) & (0.060) & (0.100) \\ (0.051) & (0.060) & (0.115) \\ (0.072) & (0.073) & (0.073) \\ (0.073) & (0.073) & (0.073) \\ (0.073) & (0.073) & (0.073) \\ (0.073) & (0.073) & (0.073) \\ (0.073) & (0.073) & (0.073) \\ (0.073) & (0.073) & (0.073) \\ (0.060) & (0.115) \\ (0.073) & (0.073) & (0.073) \\ (0.060) & (0.115) \\ (0.073) & (0.073) & (0.073) \\ (0.060) & (0.115) \\ (0.060) & (0.115) \\ (0.060) & (0.115) \\ (0.060) & (0.100) \\ (0.100) & (0.100) \\ (0.060) & (0.100) \\ (0.100) & (0.1$	Objective to promote things in common	-0.107	-0.112	-0.006
$ \begin{array}{c cccc} & & & & & & & & & & & & & & & & & $	Cultural diversity in teacher training	(0.072) 0.094^*	0.094	0.031
Class FEYesYesYesStudent controlsYesYesYesTeacher controlsNoYesYesStudent FENoNoYesAdjusted R^2 0.3940.3930.653Observations2.5392.5392.539		(0.051)	(0.060)	(0.100)
Student controlsYesYesYesTeacher controlsNoYesYesStudent FENoNoYesAdjusted R^2 0.3940.3930.653Observations2.5392.5392.539	Class FE	Yes	Yes	Yes
No Yes Yes Student FE No No Yes Adjusted R ² 0.394 0.393 0.653 Observations 2.539 2.539 2.539	Student controls	Yes	Yes	Yes
Adjusted R ² 0.394 0.393 0.653 Observations 2.539 2.539 2.539	Student FE	No	No	Yes
Adjusted K ⁻ 0.394 0.393 0.653 Observations 2.539 2.539 2.539	A limited D2	0.904	0.000	0.052
	Adjusted K ⁻ Observations	$0.394 \\ 2.539$	0.393 2.539	0.653 2.539

Table A19: Effect of Foreign Origin Teacher on ReadingComprehension Controlling For Teacher's Cultural Orientation

Panel A: All Students	(1)	(2)	(3)
Foreign origin teacher	0.183^{***}	0.230***	0.246^{*}
	(0.069)	(0.066)	(0.128)
Foreign origin teacher * teacher grade	-0.068	-0.157^{*}	-0.207^{*}
	(0.074)	(0.094)	(0.112)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.377	0.378	0.661
Observations	7,351	$7,\!351$	7,351
Panel B: Native Students			
Foreign origin teacher	0.249**	0.299***	0.244
	(0.111)	(0.105)	(0.204)
Foreign origin teacher * teacher grade	-0.098^{\prime}	-0.165	-0.269^{*}
	(0.087)	(0.119)	(0.158)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.362	0.363	0.648
Observations	4,812	4,812	4,812
Panel C: Foreign Origin Students			
Foreign origin teacher	0.144	0.130	0.375*
0 0	(0.088)	(0.096)	(0.203)
Foreign origin teacher * teacher grade	0.068	-0.021	-0.086
	(0.125)	(0.134)	(0.226)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.394	0.393	0.655
Observations	2,539	2,539	2,539

Table A20: Effect of Foreign Origin Teacher on ReadingComprehension Controlling For Grade Interaction

Panel A: All Students	(1)	(2)	(3)
Foreign origin teacher	0.122^{*}	0.144^{**}	0.238*
	(0.073)	(0.073)	(0.136)
Math test score	0.476^{***}	0.475^{***}	0.164^{***}
	(0.013)	(0.013)	(0.033)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.486	0.485	0.666
Observations	7,346	7,346	7,346
Panel B: Native Students			
Foreign origin teacher	0.197	0.217^{*}	0.262
rorongii origini teatener	(0.120)	(0.119)	(0.221)
Math test score	0.470***	0.468***	0.175***
	(0.017)	(0.017)	(0.040)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.471	0.471	0.655
Observations	4,811	4,811	4,811
Panel C: Foreign Origin Students			
Foreign origin teacher	0.079	0.103	0.347*
0 0	(0.075)	(0.088)	(0.195)
Math test score	0.457***	0.454***	0.133^{*}
	(0.027)	(0.028)	(0.074)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.490	0.488	0.659
Observations	2,535	2,535	2,535

Table A21: Effect of Foreign Origin Teacher on ReadingComprehension Controlling For Math Test Score

Panel A: All Students	(1)	(2)	(3)
Foreign origin teacher	0.241^{***}	0.155^{**}	0.221^{**}
	(0.073)	(0.062)	(0.093)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.202	0.203	0.609
Observations	$3,\!457$	3,457	$3,\!457$
Panel B: Native Students			
Foreign origin teacher	0.223	0.072	0.012
	(0.139)	(0.114)	(0.184)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R^2	0.200	0.203	0.605
Observations	2,329	2,329	2,329
Panel C: Foreign Origin Students			
Foreign origin teacher	0.305^{***}	0.015	0.394^{*}
0 0	(0.105)	(0.147)	(0.231)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted R ²	0.205	0.204	0.598
Observations	$1,\!128$	$1,\!128$	$1,\!128$

Table A22: Effect of Foreign Origin Teacher onReading Comprehension – Academic Track

Panel A: All Students	(1)	(2)	(3)
Foreign origin teacher	0.151	0.301**	0.297
	(0.099)	(0.128)	(0.212)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.272	0.273	0.568
Observations	$3,\!894$	$3,\!894$	3,894
Panel B: Native Students			
Foreign origin teacher	0.355^{**}	0.651^{***}	0.503**
	(0.151)	(0.192)	(0.244)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.255	0.256	0.558
Observations	$2,\!483$	2,483	2,483
Panel C: Foreign Origin Students			
Foreign origin teacher	-0.028	0.034	0.301
	(0.102)	(0.144)	(0.344)
Class FE	Yes	Yes	Yes
Student controls	Yes	Yes	Yes
Teacher controls	No	Yes	Yes
Student FE	No	No	Yes
Adjusted \mathbb{R}^2	0.290	0.285	0.550
Observations	1,411	1,411	1,411

Table A23: Effect of Foreign Origin Teacher onReading Comprehension – Non-Academic Track